%Title: Flavour Singlet ${\cal O}(\alpha_s^2 m_b^2/m_t^2)$ Corrections to the Partial Decay Rate $\Gamma(Z\rightarrow b\bar{b})$
%Authors: K.G. Chetyrkin and A. Kwiatkowski 
%Published: <EM> Phys. Lett. </EM> <B> B319 </B> (1993) 307-312. 
%hep-ph/9310229
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% eps files appended
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\documentstyle[11pt,epsfig]{article}
\setlength{\textwidth}{14.5cm}
\setlength{\textheight}{22.5cm}
\setlength{\oddsidemargin}{1.cm}
\setlength{\evensidemargin}{1.cm}
\setlength{\headheight}{0.5cm}
\setlength{\headsep}{1.0cm}
\setlength{\topmargin}{0.0cm}
\setlength{\topskip}{0.1cm}
\setlength{\footheight}{0.5cm}
\renewcommand{\baselinestretch}{1.1}
\setlength{\footskip}{1.5cm}
\frenchspacing
\newcommand{\beq}{\begin{equation}}
\newcommand{\eeq}{\end{equation}}
\newcommand{\ba}{\begin{array}}
\newcommand{\ea}{\end{array}}
\newcommand{\EQN}{\label}
\newcommand{\dsp}{\displaystyle}
\newcommand{\api}{\frac{\alpha_s}{\pi}}
\def\bbuildrel#1_#2^#3%
  {\mathrel{\mathop{\kern 0pt#1}\limits_{#2}^{#3}}}
                             



\begin{document}
\begin{thebibliography}{99}

\bibitem{KniKue90} 
B.A.Kniehl, J.H.K\"uhn, 
Phys.\ Lett.\ B 224 (1990) 229.\\
B.A.Kniehl, J.H.K\"uhn,
 Nucl.\ Phys.\ B 329 (1990) 547.

\bibitem{CheKwi93}
K.G.Chetyrkin, A.Kwiatkowski,
 Phys.\ Lett.\ B 305 (1993) 285. 

\bibitem{Sch93}
D.Schaile, talk given at the Ringberg
 Workshop 1993: Electroweak Results 
at LEP.

\bibitem{CheKue90}
K.G.Chetyrkin, J.H.K\"uhn, Phys.\ Lett.\  
248 B (1990) 359.

\bibitem{CheKueKwi92} 
K.G.Chetyrkin, J.H.K\"uhn, A.Kwiatkowski,
 Phys.\   Lett. B 282 (1992) 221.  

\bibitem{Mni93}
J.Mnich, talk given at the XIII International 
Conference on Physics in 
Collision, 17 June 1993, Heidelberg:
 Precision Tests of Electroweak Theory at LEP. 

\bibitem{Hea91}
A.C.Hearn, REDUCE user's manual, Version 3.4,
 Rand publication CP78 (1991).

\bibitem{LarTkaVer91}
S.A.Larin, F.V.Tkachov,J.A.M.Vermaseren,
Preprint NIKHEF-H/91-18.

\bibitem{Ver91}
J.A.M.Vermaseren, Symbolic Manipulation
 with FORM, Version 2, CAN, 
Am\-ster\-dam 1991.

\bibitem{HooVel72}
G.'t Hooft, M.Veltman,
 Nucl.\ Phys.\ B 44 (1972) 189.   


\bibitem{BreMai77}
P.Breitenlohner, D.Maison,  
Comm. Math. Phys.  
52  (1977) 11.

\bibitem{NacWet79}
O.Nachtmann, W.Wetzel, 
Phys.\ Lett.\ 81 B (1979) 211.        

\bibitem{PivTka84}
G.B.Pivovarov, F.V.Tkachov,
 Preprint INR P-0370 (1984), Moscow; \\
F.V.Tkachov, Int.\ Journ.\ Mod.\ Phys. 
A 8 (1993) 2047; \\
G.B.Pivovarov, F.V.Tkachov,  Int.\ Journ.\
 Mod.\ Phys. A 8 (1993) 2241.

\bibitem{GorLar87}
S.G.Gorishny, S.A.Larin, Nucl.\ Phys. B 287 (1987) 452;\\ 
S.G.Gorishny, Nucl.\ Phys. B 319 (1989) 633.

\bibitem{CheSmi87}
K.G.Chetyrkin,V.A.Smirnov,
 Preprint INR P-518 (1987), Moscow; \\
K.G.Chetyrkin, Preprint MPI-PAE/PTh 13/91 (1991), 
Munich; \\
V.A.Smirnov, Commun.\ Math.\ Phys. 134 (1990) 109. 

\bibitem{Smi91}
V.A.Smirnov, Renormalization and Asymptotic Expansion,
(Birkh\"auser, Basel, 1991).

\bibitem{CheKue93}
K.G.Chetyrkin, J.H.K\"uhn,Phys.\ Lett.\ B 308 (1993) 127.

\bibitem{DomPav92}
C.A.Dominguez, N.Paver, Phys.\ Lett. B 293 (1992) 197.

\bibitem{m0}
K.G.Chetyrkin, A.L.Kataev, F.V.Tkachov, 
Phys.\  Lett.\ B 85 (1979) 277; \\
M.Dine, J.Sapirstein,
 Phys.\ Rev.\ Lett.\ 43 (1979) 668; \\
W.Celmaster, R.J.Gonsalves, 
Phys.\ Rev.\ Lett.\ 44 (1980) 560.

\bibitem{m1}
S.G.Gorishny, A.L.Kataev, S.A.Larin,
Phys.\ Lett.\ B 259 (1991) 144; \\
L.R.Surguladze, M.A.Samuel, 
Phys.\ Rev.\ Lett.\ 66 (1991) 560.

\bibitem{Che93}
K.G.Chetyrkin, Phys.\ Lett.\ B 307 (1993) 169.

\end{thebibliography}

\begin{titlepage}
\noindent
%
%
\vspace{0.5cm}
\begin{center}
  \begin{Large}
  \begin{bf}
Flavour Singlet ${\cal O}(\alpha_s^2 m_b^2/m_t^2)$
 Corrections to the
Partial Decay Rate $\Gamma(Z\rightarrow b\bar{b})$
   \\
  \end{bf}
  \end{Large}
%
% Author
%
  \vspace{0.8cm}
  \begin{large}
K.G.Chetyrkin    \\[3mm]
    Institute for Nuclear Research \\
    Russian Academy of Sciences\\ 
    60th October Anniversary Prospect 7a\\[2mm]
    Moscow 117312, Russia\\[8mm]
    A.Kwiatkowski    \\[3mm]
    Institut f\"ur Theoretische Teilchenphysik\\
    Universit\"at Karlsruhe\\[2mm]
    76128 Karlsruhe, Germany\\
  \end{large}
%
% Abstract
%
  \vspace{3cm}
  {\bf Abstract}
\end{center}
\begin{quotation}
\noindent
We analytically compute the flavour singlet
 ${\cal O}(\alpha_s^2 m_b^2/m_t^2)$
radiative corrections to the partial decay rate
 $\Gamma(Z\rightarrow b\bar{b})$.
These corrections arise from  anomalous
``double triangle'' diagrams 
containing a  single $\gamma_5$ 
matrix inside each of two closed fermion loops.
They represent the next-to-leading
 term of the asymptotic expansion in the
inverse large top mass. 
 As a byproduct of the calculcation
we confirm the results of B.Kniehl and J.H.K\"uhn  
for the flavour singlet massless corrections
 \cite{KniKue90}
 as well as our previous
${\cal O}(\alpha_s^2 m_b^2/M_Z^2)$ result 
\cite{CheKwi93}.

\end{quotation}
\end{titlepage}

\section{Introduction}
\renewcommand{\arraystretch}{2}                             %

The decay of the $Z$-boson into a bottom
 quark pair has become of 
increasing interest in the past. On the 
experimental side the measurement
of the partial decay rate $\Gamma(Z\rightarrow
 b\bar{b})$
represents one of the high precision
 tests of the Standard Model, which
are performed at LEP. The experimental
 uncertainty of this quantity
 $\Gamma(Z\rightarrow b\bar{b}) = 
(383\pm 5.6)$ MeV is likely
 to be reduced even below 1\% in the future 
\cite{Sch93}. 

In order to  match this accuracy
 also on the theoretical side, a 
precise knowledge of radiative corrections
 to the partial $Z$-decay rate becomes mandatory.
For this reason the influence of the
 finite bottom mass became a relevant
 topic in QCD calculations. Bottom mass 
effects have conveniently been 
studied by expanding radiative corrections 
in $m_b^2/M_Z^2$, which is 
certainly a small parameter.
 Several contributions
of the quadratic mass corrections are known
 in the literature.
Flavour non-singlet corrections were obtained
 to third order
${\cal O}(\alpha_s^3 m_b^2/M_Z^2)$ for the vector 
current induced rate
\cite{CheKue90} and to second order ${\cal O}
(\alpha_s^2 m_b^2/M_Z^2)$
 to the axial vector induced rate
 \cite{CheKueKwi92}.
Flavour singlet contributions to the $Z$-width 
were first discussed 
for the massless case in \cite{KniKue90}. It was
 pointed out that
they arise at second order only for the axial
 induced rate, whereas
the vector contribution vanishes at this order 
due to Furry's theorem.
The corresponding mass corrections were obtained 
in \cite{CheKwi93}.
 
In this paper we focus again on the flavour singlet
 corrections to the partial $Z$-width. The
accurate  measurement of the various free  
parameters of the Standard Model made it possible 
to derive constraints on the 
top mass. With $m_t=(158^{+30}_{-34}\pm 20 \pm 12)$ GeV 
\cite{Mni93}
the top quark is expected to be roughly
twice as heavy as the $Z$-boson. Therefore one 
can expect mass corrections
of order ${\cal O}(m_b^2/m_t^2)$ to be of similar 
size as those of order
${\cal O}(m_b^2/M_Z^2)$.
 The inclusion
of the ${\cal O}(m_b^2/m_t^2)$ contribution 
 completes the
 discussion about flavour singlet bottom mass 
corrections.

Our approach for the actual calculation
of the decay rate
\beq \EQN{i1}
\Gamma^A(Z\rightarrow b\bar{b})
=\frac{G_FM_Z^3}{24\sqrt{2}\pi}N_Ca_b^2R^A
\eeq
is to employ its relation to the 
absorptive part of 
the corresponding axial current 
correlation function
\beq \EQN{i1a}
R^A(s) = -\frac{4\pi}{s} 
\mbox{\rm Im} \Pi^A_1(s+i\epsilon)
.\eeq
Here $\Pi^A_1$ arises from the Lorentz 
decomposition of the 2-point
current correlator 
\beq \EQN{i2}
\ba{ll}\dsp
\Pi^A_{\mu\nu}(q)
& \dsp
=i\int dx e^{iqx}\langle 0|\;T\;\Delta_{\mu}(x)
\Delta_{\nu}(0)\;|0\rangle
\\ & \dsp
= g_{\mu\nu}\Pi^A_1(q^2) + q_{\mu}q_{\nu}\Pi^A_2(q^2)
\ea\eeq
with $\Delta_{\mu} = A^t_{\mu}-A^b_{\mu}$,
 where we denote the axial vector current for the 
quark of flavour $f$ with $A^f_{\mu} =
 \bar{\Psi}_f\gamma_{\mu}\gamma_5\Psi_f$.

In order to extract the $m_b^2/m_t^2$-contribution
 of the correction, we
integrate out
the heavy top quark from the
theory. This results in a power expansion 
with respect to the inverse 
heavy mass $1/m_t$ and leads us to the 
$m_b^2/m_t^2$-term we are 
interested in. 

As a byproduct of the calculation we  recover the 
leading and next-to-leading top dependence 
of the massless ($m_b=0$) flavour singlet
corrections of \cite{KniKue90} as well
 as the leading contribution
of our corresponding  massive ($m_b\neq 0$) calculation 
\cite{CheKwi93}.
Whereas in \cite{CheKwi93} we made use of 
the symbolic manipulation system REDUCE
\cite{Hea91}, the complexity of the calculation 
now exceeds the capability
 of this software. In this work our computation
 is performed 
with the help of the program MINCER \cite{LarTkaVer91} 
which is designed for 
massless 
multiloop calculations and is based on the 
symbolic manipulation package FORM
\cite{Ver91}.

This paper is organized in the following way.
In Section 2 we give a description of the
 calculation of the 
${\cal O}(\alpha_s^2 m_b^2/m_t^2)$-correction. 
We briefly address our  treatment of $\gamma_5$
 in $D\neq 4$ dimensions. 
Further emphasis is laid on the details of the
 hard mass procedure. 
Afterwards
the verification of the existing lower 
order results is presented.
 The numerical discussion and conclusions 
are the contents of
Section 3.  
   
\section{Description of the Calculation}

The two-point correlation function $\Pi^A_{\mu\nu}$ 
is represented
 graphically by propagator-type ``light-by-light''
 Feynman diagrams,
which are characterized by two closed fermion loops
 and a purely gluonic
intermediate state. We need not consider 
the diagonal pieces of
 $\Pi^A_{\mu\nu}$: The diagram with two top quark
 loops may be disregarded,
because the $Z$-boson cannot decay into top quarks
 due to the heaviness of 
the top.
Furthermore, being interested in ${\cal O}
(\alpha_s^2 m_b^2/ m_t^2)$
corrections to $\Gamma(Z\rightarrow 
b\bar{b})$, also the
``light'' graph (see Figure 1a) with two bottom
loops does not contribute to this order.
 We therefore have to calculate the 
``heavy'' diagrams
of Figure 1b. They carry a relative factor $2$
 as compared to the 
light diagram due to the top-bottom symmetry.  
 The double arrows at the fermion 
lines indicate the
two different diagrams with opposite 
loop orientation. 
%##################
\begin {figure}[t]                                        
\begin {tabular}{ccrc}  
\parbox{3cm}{   
\epsfig{file=gl.eps,width=5.cm,height=5.cm} 
}                                                                   
&\hphantom{XXX}  & $2\times$ &
\parbox{3cm}{
\epsfig{file=ghar.eps,width=5.cm,height=5.cm}
            }
\end {tabular}      
\caption { Flavour singlet diagrams contributing to
 $\Gamma(Z\rightarrow b\bar{b})$
 (thin lines: bottom, thick lines: top).} 
\end{figure}   
%###########################
The two-point function $\Pi^A_{\mu\nu}$ of 
eq.(\ref{i2}) contains the 
difference between the axial currents of the 
top and the bottom quark
and is therefore an anomaly free quantity. 
Its absorptive part
 is also ultraviolet (UV)
finite. Although the contributions
 of the separate diagrams 
are UV divergent, their singularities are 
independent of the fermion mass
and compensate each other. 
 At intermediate stages of the calculation
 the introduction of a regularization
procedure  therefore becomes necessary
since we calculate 
the  light and the heavy diagrams separately.

 Concerning the  question of how to
define the $\gamma_5$-matrix in $D\neq 
4$ dimensions, we essentially
use the definition of 't Hooft and Veltman 
\cite{HooVel72}
formalized by Breitenlohner and Maison \cite{BreMai77}.
 In $D= 4-2\epsilon$ dimensions we will work
 with the generalized current
\beq\EQN{d1}
A_i^{[\mu\nu\lambda]} = \bar{\Psi}_i 
\gamma^{[\mu\nu\lambda]} \Psi_i
,\eeq
where 
$\gamma^{[\mu\nu\lambda]}  
 = ( \gamma^{\mu}\gamma^{\nu}\gamma^{\lambda} 
- \gamma^{\lambda}\gamma^{\nu}\gamma^{\mu})/2
$
and
$i=b,t$.
Taking the limit $D\rightarrow 4$ at the end
 of the calculation, when all
 divergences have been removed, we recover the
 result for the axial current
$A^{\rho}$ through
\beq\EQN{d2}
 A_{\rho} = \frac{i}{3!}\epsilon_{\rho\mu\nu\lambda}
             A^{[\mu\nu\lambda]} 
.\eeq
The corresponding generalized current correlator
may be decomposed in the following way
\cite{NacWet79}
\beq \EQN{d3}
\ba{ll} \Pi^{[\mu\nu\lambda]}_{[\mu'\nu'\lambda']} 
& = \dsp i \int dx e^{iqx} 
\langle 0|\;
T\; A^{[\mu\nu\lambda]}(x)
A^{[\mu'\nu'\lambda']}(0)\; |0 \rangle 
\\ &
 = \dsp {[\mu\nu\lambda] \atop [\mu'\nu'\lambda']}
{\rm P}_1(q^2)
+g^{[\mu}_{[\mu'}g^{\nu}_{\nu'}
q^{\lambda ]}q_{\lambda']}  
   {\rm P}_2(q^2)  ,  \ea 
\eeq
where 
$
 {[\mu\nu\lambda] \atop [\mu'\nu'\lambda']}    
= \frac{1}{3!} {\rm det}(g_{\alpha\alpha'})$ with
$
\alpha=\mu,\nu,\lambda
$
and
$ 
 \alpha'=\mu',\nu',\lambda'
$ 
and where $g^{[\mu}_{[\mu'}g^{\nu}_{\nu'}
q^{\lambda ]}q_{\lambda']}$
is  fully antisymmetrized with respect 
to $[\mu\nu\lambda]$
and $[\mu'\nu'\lambda']$ separately.
It is  convenient to work with the 
contracted tensors
$
P_L=\Pi^{[\mu\nu\lambda]}_{[\mu\nu\lambda]}   
$
and 
$
P_Q= q^{\lambda}q^{\lambda'}
\Pi^{[\mu\nu\lambda]}_{[\mu\nu\lambda']}  
$.
It turns out that the spectral function 
$\Pi^A_1$, which enters
the $Z$-decay rate, is already completely
 determined by $P_Q$:
\beq\EQN{d4}
\Pi^A_1 =  \frac{1}{6}
\frac{P_Q}{q^2}
.\eeq 

The calculation of the heavy diagrams of 
Figure 1 proceeds as follows.
We apply a hard mass expansion in the
 inverse heavy top mass $1/m_t$.
In practice this means that the Feynman
 integral $\langle\Gamma \rangle
(q^2)$ corresponding to a Feynman diagram 
$\Gamma$ is represented by an 
asymptotic expansion 
 \cite{PivTka84,GorLar87,CheSmi87,Smi91}):
\beq
\EQN{d5}
<\Gamma>(q^2)
\bbuildrel{=\!=\!=}_{{\scriptstyle{m_t\to\infty}}}^{}                       
\sum_\gamma  C^{(t)}_\gamma \star 
\langle\Gamma/\gamma\rangle^{\mbox{\rm
 \scriptsize eff}} 
.\eeq
Thereby the hard subgraphs $\gamma$
 of $\langle \Gamma\rangle$ are reduced to 
effective vertices such that  the diagram   
$\langle\Gamma/\gamma\rangle^{\mbox{\rm 
\scriptsize eff}}$
is built up only from the (light) 
particles of the effective theory.
The subintegrals $C^{(t)}_\gamma$, 
which contain the whole top dependence, are
expanded in a formal (multidimensional)
 Taylor series with respect
to their external momenta and the light 
bottom mass. This Taylor
expansion is then inserted into the effective 
vertex and the whole 
expression is evaluated.

In our case we can identify two possible 
hard subgraphs in our diagrams
as is shown in Figure 2.
%########################
\begin {figure}[t]                                        
\begin {tabular}{ccccc}  
\parbox{3cm}{   
\epsfig{file=gh.eps,width=4.cm,height=4.cm} 
}                                                                   
& $\longrightarrow$ &
\parbox{3cm}{
\epsfig{file=ct1.eps,width=4.cm,height=4.cm}
            }
& $\star$ &
\parbox{3cm}{ 
\epsfig{file=gh1.eps,width=4.cm,height=4.cm} 
            }  
\\    
& $+$ & 
\parbox{3cm}{
\epsfig{file=ct2.eps,width=4.cm,height=4.cm}  
            }
& $\star$ &
\parbox{3cm}{ 
\epsfig{file=gh2.eps,width=4.cm,height=4.cm}  
       }  
\end {tabular}      
\caption {The hard mass procedure.}
\end{figure}   
%################
The hard mass procedure results in a
 power series in the inverse top mass.
The leading order of this expansion
 has the top mass dimension $m_t^0$
and receives its contribution only
 from the two-loop subgraph $C^{(t)}_1$.
For the next-to-leading order $\sim
 1/m_t^2$ the second order terms in the 
Taylor expansion of $C^{(t)}_1$ as 
well as the  contributions of the
 one-loop subgraph $C^{(t)}_2$ are 
needed. Since the Taylor series
is evaluated at vanishing external momenta, 
the subgraphs are computed by
 solving two- and one-loop massive
tadpole integrals respectively. 

After having done this, the problem is 
reduced to the calculation of 
one- and two-loop diagrams, which still 
contain the bottom quark mass.
We perform a subsequent expansion
 in $m_b^2$.
Picking out the quadratic mass terms 
in order to obtain the 
order ${\cal O}(\alpha_s^2 m_b^2/ m_t^2)$ 
corrections to the $Z$-decay rate,
we are left with massless  propagator
 type diagrams, which we compute 
 with the help
of the multiloop program MINCER.
For the order ${\cal O}
(\alpha_s^2 m_b^2/m_t^2)$ corrections
we arrive at the following result:
\beq\EQN{d6}
R^A = - 10 \left( \frac{\alpha_s}{\pi}\right)^2
\frac{\bar{m}_b^2}{m_t^2}\left[ \frac{8}{81}-
\frac{1}{54}\ln\frac{s}{m_t^2}
\right]
\eeq


\vspace{2ex}
Before closing this section let us add 
some additional details and remarks.
In deriving the result of eq.(\ref{d6})
 we have isolated the next-to-leading
term in the heavy mass expansion and
 also concentrated on the 
quadratic contribution of the bottom
 mass. As a byproduct of this 
calculation we automatically obtain 
also the lower order pieces
with respect to the $m_b^2$-expansion
 as well as the $1/m_t$-power
series. Therefore we are able to
 compare the outcome of the 
calculation not only with our
 previous result for the 
quadratic mass corrections for
 the flavour singlet diagrams
\cite{CheKwi93}, but also
with the result of \cite{KniKue90}
 for the massless case.
In order to verify the latter
 corrections, we have to extend our consideration 
and include the light diagram of
 Figure 1. For this diagram we  perform 
an expansion in $m_b^2$ and compute
 the resulting massless 
three-loop integrals with the program MINCER.
We present the intermediate results
for the absorptive parts  of the
 various contributions, 
namely of the light diagram $\Gamma_\ell$ 
and of the hard diagrams, separated 
according to
the two-loop hard subgraph 
($\Gamma_{h1}$) and the one-loop hard
subgraph ($\Gamma_{h2}$):
\beq\EQN{d7}
\ba{ll}\dsp
R^A(\Gamma_\ell) =
& \dsp
\left( -\frac{1}{3}\right)\left( 
\frac{\alpha_s}{\pi}\right)^2
 \left[ \frac{3}{2}\frac{1}{\epsilon}
+\frac{25}{2} - \frac{9}{2}\ln\frac{s}
{\mu^2}\right] 
\\ \dsp
& \dsp
 -6\frac{\bar{m}_b^2}{s}\left( 
\frac{\alpha_s}{\pi}\right)^2
 \left[ -\frac{1}{2}\frac{1}{\epsilon}
-\frac{15}{4} + \frac{3}{2}\ln\frac{s}
{\mu^2}\right] 
\ea\eeq
\beq\EQN{d8}
\ba{ll}\dsp
R^A(\Gamma_{h1}) =
& \dsp
 \left(-\frac{1}{3}\right)
\left( \frac{\alpha_s}{\pi}\right)^2
 \left[- \frac{3}{2}\frac{1}{\epsilon}
-\frac{13}{4} +\frac{3}{2}\ln\frac{s}{\mu^2}
- 3\ln\frac{\mu^2}{m_t^2}
 -\frac{10}{27}\frac{s}{m_t^2}
\right] 
\\ \dsp
& \dsp
 -6\frac{\bar{m}_b^2}{s}\left(
 \frac{\alpha_s}{\pi}\right)^2
 \left[ \frac{1}{2}\frac{1}{\epsilon}
+\frac{3}{4} - \frac{1}{2}
\ln\frac{s}{\mu^2}
+\ln\frac{\mu^2}{m_t^2}
\right]
\\ \dsp
& \dsp
 -10\frac{\bar{m}_b^2}{m_t^2}\left(
 \frac{\alpha_s}{\pi}\right)^2
 \left[ -\frac{1}{54}\frac{1}{\epsilon}
+\frac{7}{324} + \frac{1}{54}
\ln\frac{s}{\mu^2}
- \frac{1}{27}\ln\frac{\mu^2}{m_t^2}
\right] 
\ea \eeq
\beq\EQN{d9}
\ba{ll}\dsp
R^A(\Gamma_{h2}) =
& \dsp
 \left(-\frac{1}{3}\right)\left( 
\frac{\alpha_s}{\pi}\right)^2
 \frac{1}{9}
\frac{s}{m_t^2}
\\ \dsp
& \dsp
 -10\frac{\bar{m}_b^2}{m_t^2}
\left( \frac{\alpha_s}{\pi}\right)^2
 \left[ \frac{1}{54}\frac{1}{\epsilon}
+\frac{25}{324} -
 \frac{1}{27}\ln\frac{s}{\mu^2}
+\frac{1}{54}\ln\frac{\mu^2}{m_t^2}
\right] 
\ea \eeq
One observes that the pole terms of the 
leading order cancel in the sum of
the  diagrams $\Gamma_\ell$ and 
$\Gamma_{h1}$, whereas those of the
 next-to-leading contribution compensate
 each other in the combination 
of $\Gamma_{h1}$ and 
$\Gamma_{h2}$. In a similar way
the $\mu^2$-dependence vanishes 
in the sum of the logarithms.
Finally  the result is 
gauge invariant. The independence of 
each of the three contributions
on the QCD gauge parameter serves as 
a further check of the calculation. 
We arrive at the following  result:
\beq\EQN{d10}
\ba{ll} \dsp
R^A_{singlet} = 
& \dsp
\left(-\frac{1}{3}\right)\left(
\frac{\alpha_s}{\pi}\right)^2
\left[ \frac{37}{4} - 3 \ln\frac{s}{m_t^2} 
- \frac{7}{27}\frac{s}{m_t^2}
\right]
\\ & \dsp
 -6\frac{\bar{m}_b^2}{s}\left(
 \frac{\alpha_s}{\pi}\right)^2
 \left[ -3+ \ln\frac{s}{m_t^2}
\right]
\\ 
& \dsp
 -10\frac{\bar{m}_b^2}{m_t^2}\left( 
\frac{\alpha_s}{\pi}\right)^2
 \left[
\frac{8}{81} - \frac{1}{54}\ln\frac{s}{m_t^2}
\right] 
.\ea\eeq
The first line of this equation 
reproduces the result of 
\cite{KniKue90}.
 It  was also calculated up to order 
${\cal O}(\alpha_s^2s^2/m_t^4)$ in \cite{CheKue93}. 
The second line verifies 
our previous calculation in 
\cite{CheKwi93}, and the third line
represents the new contribution.
 
\section{Discussion}
Let us first discuss
 the numerical size of our result for 
the flavour singlet ${\cal O}(
\alpha_s^2 m_b^2/m_t^2)$ corrections to the 
partial decay rate of the $Z$-boson.
Collecting the complete mass corrections
 for the axial vector induced
rate we obtain 
\beq\EQN{r1}
\ba{ll} \dsp
R^A = 
& \dsp 
1 - 6 \frac{\bar{m}_b^2}{s}
\left( 1 + \frac{11}{3}\left( 
\frac{\alpha_s}{\pi}\right)
+\left( \frac{\alpha_s}{\pi}
\right)^2\left[ 11.286 
+ \ln\frac{s}{m_t^2}\right]
\right)
\\ &\dsp
\hphantom{1} - 10 \frac{\bar{m}_b^2}{m_t^2}
\left( \frac{\alpha_s}{\pi}\right)^2\left[ 
\frac{8}{81} 
-\frac{1}{54} \ln\frac{s}{m_t^2}\right]
.\ea\eeq
We compare the size of the corrections 
of  the orders ${\cal O}(\alpha_s^2 m_b^2/M_Z^2)$ 
and ${\cal O}(\alpha_s^2 m_b^2/m_t^2)$
  in Table 1
for different $m_t$. The numbers are based
 on $\Lambda_{QCD}=400$ MeV and
an on-shell mass of $m_b=4.72$ Gev 
\cite{DomPav92}, which corresponds to 
$\alpha_s(M_Z^2)= 0.131$ and
 $\bar{m}_b(M_Z^2)=2.61$ GeV.   
One can observe that
the order ${\cal O}(\alpha_s^2 m_b^2/m_t^2)$ 
corrections carry opposite
sign as the ${\cal O}(\alpha_s^2 m_b^2/M_Z^2)$
 flavour singlet corrections.
In size  they are smaller about one order
of magnitude.  
Due to their polynomial $1/m_t^2$ behavior 
they are much more sensitive 
on the top mass. However, 
the complete flavour singlet mass  corrections
 amount to only  $\sim 0.03$ permille
and are too small 
for LEP to be sensitive
 on the top mass
from flavour singlet QCD effects. 
 
\begin{table}
\renewcommand{\arraystretch}{1}                             %
{\bf Table 1:} Flavour singlet mass 
corrections of  $R^A.$ 
 
\vspace{2ex}

\begin{tabular}{|c||c|c||c|}\hline  
$m_t$ & $R^A(\bar{m}_b^2/M_Z^2)$ & $R^A
(\bar{m}_b^2/m_t^2)$ & $\sum$
\\ 
in GeV & $\cdot 10^5$ & $\cdot 10^5$ 
& $\cdot 10^5$ \\ 
    \hline
110 & 2.87 & -0.103 & 2.76 \\
130 & 3.16 & -0.078 & 3.08 \\
150 & 3.40 & -0.061 & 3.33 \\
170 & 3.61 & -0.050 & 3.56 \\ \hline
\end{tabular} 
\renewcommand{\arraystretch}{2}                             %
\vspace{2ex}
\end{table}

Together with
 the flavour non-singlet contribution the 
second order mass corrections are just 
below $-0.1$ permille in magnitude
(see column 6 of Table 2).
The size of the  quadratic mass
 corrections originating from 
the separate orders in the coupling
 constant are presented in
Table 2 as well.  


\begin{table}
\renewcommand{\arraystretch}{1}                             %
{\bf Table 2:} Quadratic 
 mass corrections of $R^A.$  

\vspace{2ex}

\begin{tabular}{|c|c|c||c|c|c||c|}\hline  
$\Lambda_{QCD}$ & $\alpha_s(M_Z^2)$ 
& $\bar{m}_b(M_Z^2)$ & 
$R^A(\alpha_s^0 \bar{m}_b^2)$ & $R^A(\alpha_s \bar{m}_b^2)$ 
& $R^A(\alpha_s^2 \bar{m}_b^2)$ &
$\sum$ 
\\ 
in MeV & & in GeV & $\cdot 10^3$  & 
$\cdot 10^3$ & $\cdot 10^3$ & $\cdot 10^3$
 \\     \hline
200 & 0.117 & 2.94 & -6.24 & -0.85 & -0.092 & -7.34 \\
400 & 0.131 & 2.61 & -4.90 & -0.74 & -0.090 & -5.74 \\
650 & 0.143 & 2.27 & -3.70 & -0.62 & -0.081 & -4.40 \\
\hline
\end{tabular} 

\renewcommand{\arraystretch}{2}                             %
\vspace{2ex}
\end{table}

For completeness we finally reproduce the 
complete formula including all presently known 
QCD corrections to the partial $Z$-decay rate.
\beq   \EQN{r2}
\ba{ll}
\dsp \Gamma(Z\rightarrow b\bar{b}) 
& \dsp =  \Gamma^V + \Gamma^A 
\\ & \dsp = \frac{G_F M_Z^3}{24\sqrt{2}\pi}
         N_C (v_b^2 R^V + a_b^2 R^A),
\ea
\eeq
where
\beq  \EQN{r3}
\ba{ll}
R^V & \dsp = 1 + \api + 
\left[ 1.41 - \frac{8}{135}r\ln(4r) + 
\frac{176}{675}r \right]
 \left(\api\right)^2
\\ & \dsp    
 - 12.76 \left(\api\right)^3    
-\frac{\sum_f v_f}{3 v_b} 1.24 \left(\api\right)^3
\\ & \dsp
+ 12 \frac{\bar{m}_b^2}{s} \api \left\{ 1 + 8.7 \api 
           + 45.15  \left(\api\right)^2  \right\}
\\ \dsp
R^A & \dsp = 1 + \api + 
\left[ 1.41 - \frac{8}{135}r\ln(4r) + 
\frac{176}{675}r 
+ \frac{1}{3}{\cal I}(r)
 \right] \left(\api\right)^2    
\\ & \dsp
- 6 \frac{\bar{m}_b^2}{s}  \left\{ 1 + \frac{11}{3} \api 
           + \left[ 11.286 + \ln \frac{M_Z^2}{m_t^2} \right] 
\left(\api\right)^2  \right\}.
\\ &\dsp
 - 10 \frac{\bar{m}_b^2}{m_t^2}
\left( \frac{\alpha_s}{\pi}\right)^2\left\{ 
\frac{8}{81} 
-\frac{1}{54} \ln\frac{s}{m_t^2}\right\}
.\ea
\eeq
Here we have included the massless \cite{m0,m1} as well
as the quadratic mass corrections.
The term  ${\cal I}(r)$ with $r=M_Z^2/4m_t^2$
 represents the flavour singlet contribution 
of the triangle diagrams for the massless case
  \cite{KniKue90} and reads
\beq \EQN{r4}
{\cal I}(x) = -9.250 + 1.037 x + 0.632 x^2 + 6 \ln (2\sqrt{x})
.\eeq
The other terms depending on the top mass through $r$
originate from non singlet double bubble diagrams
with a virtual top loop and have been calculated 
in \cite{Che93}.
 
We finally conclude that in this work
we have determined the
 complete flavour singlet contribution
 to the quadratic mass corrections for the 
axial vector induced partial decay 
rate of the $Z$-boson
into bottom quarks. As a byproduct we
 could verify previous results
which were obtained earlier for the
 flavour singlet diagrams.

QCD corrections for the decay rate
$\Gamma(Z\rightarrow b\bar{b})$ are well established
by now. They are known  to a precision
which is sufficiently accurate in view of the
experimental uncertainities at LEP.   


{\bf Acknowledgment}

We would like to thank J.H.K\"uhn for helpful discussions.
 


\end{document}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% eps-files 
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% gl.eps %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%!PS-Adobe-2.0 EPSF-2.0
%%BoundingBox: 0 0 142 142
%%Title: /usr/users/ak/PAW/SINGLET/gl.eps
%%Creator: HIGZ Version 1.16/11
%%CreationDate: 93/10/03   12.43
%%EndComments
80 dict begin
/s {stroke} def /l {lineto} def /m {moveto} def /t {translate} def
/sw {stringwidth} def /r {rotate} def /rl {roll} def
/d {rlineto} def /rm {rmoveto} def /gr {grestore} def /f {eofill} def
/c {setrgbcolor} def /lw {setlinewidth} def /sd {setdash} def
/cl {closepath} def /sf {scalefont setfont} def
/box {m dup 0 exch d exch 0 d 0 exch neg d cl} def
/bl {box s} def /bf {box f} def
/mp {newpath /y exch def /x exch def} def
/side {[w .77 mul w .23 mul] .385 w mul sd w 0 l currentpoint t -144 r} def
/mr {mp x y w2 0 360 arc} def /m24 {mr s} def /m20 {mr f} def
/mb {mp x y w2 add m w2 neg 0 d 0 w neg d w 0 d 0 w d cl} def
/mt {mp x y w2 add m w2 neg w neg d w 0 d cl} def
/m21 {mb f} def /m25 {mb s} def /m22 {mt f} def /m26{mt s} def
/m23 {mp x y w2 sub m w2 w d w neg 0 d cl f} def
 /m27 {mp x y w2 add m w3 neg w2 neg d w3 w2 neg d w3 w2 d cl s} def
 /m28 {mp x w2 sub y w2 sub w3 add m w3 0 d 0 w3 neg d w3 0 d 0 w3 d w3 0 d
 0 w3 d w3 neg 0 d 0 w3 d w3 neg 0 d 0 w3 neg d w3 neg 0 d cl s } def
 /m29 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 4 {side} repeat cl fill gr} def
 /m30 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 5 {side} repeat s gr} def /m31 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d
 x w2 sub y w2 add m w w neg d x w2 sub y w2
 sub m w w d s} def
/m2 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d s} def
/m5 {mp x w2 sub y w2 sub m w w d x w2 sub y w2 add m w w neg d s} def
/reencdict 24 dict def /ReEncode {reencdict begin /nco&na exch def
/nfnam exch def /basefontname exch def /basefontdict basefontname findfont def
/newfont basefontdict maxlength dict def basefontdict {exch dup /FID ne
{dup /Encoding eq {exch dup length array copy newfont 3 1 roll put} {exch
newfont 3 1 roll put} ifelse} {pop pop} ifelse } forall newfont
/FontName nfnam put nco&na aload pop nco&na length 2 idiv {newfont
/Encoding get 3 1 roll put} repeat nfnam newfont definefont pop end } def
/accvec [ 176 /agrave 181 /Agrave 190 /acircumflex 192 /Acircumflex
201 /adieresis 204 /Adieresis 209 /ccedilla 210 /Ccedilla 211 /eacute
212 /Eacute 213 /egrave 214 /Egrave 215 /ecircumflex 216 /Ecircumflex
217 /edieresis 218 /Edieresis 219 /icircumflex 220 /Icircumflex
221 /idieresis 222 /Idieresis 223 /ntilde 224 /Ntilde 226 /ocircumflex
228 /Ocircumflex 229 /odieresis 230 /Odieresis 231 /ucircumflex 236 /Ucircumflex
237 /udieresis 238 /Udieresis 239 /aring 242 /Aring 243 /ydieresis
244 /Ydieresis 246 /aacute 247 /Aacute 252 /ugrave 253 /Ugrave] def
/Times-Roman /Times-Roman accvec ReEncode
/Times-Italic /Times-Italic accvec ReEncode
/Times-Bold /Times-Bold accvec ReEncode
/Times-BoldItalic /Times-BoldItalic accvec ReEncode
/Helvetica /Helvetica accvec ReEncode
/Helvetica-Oblique /Helvetica-Oblique accvec ReEncode
/Helvetica-Bold /Helvetica-Bold accvec ReEncode
/Helvetica-BoldOblique /Helvetica-BoldOblique  accvec ReEncode
/Courier /Courier accvec ReEncode
/Courier-Oblique /Courier-Oblique accvec ReEncode
/Courier-Bold /Courier-Bold accvec ReEncode
/Courier-BoldOblique /Courier-BoldOblique accvec ReEncode
/oshow {gsave [] 0 sd true charpath stroke gr} def
/stwn { /fs exch def /fn exch def /text exch def fn findfont fs sf
 text sw pop xs add /xs exch def} def
/stwb { /fs exch def /fn exch def /nbas exch def /textf exch def
textf length /tlen exch def nbas tlen gt {/nbas tlendef} if
fn findfont fs sf textf dup length nbas sub nbas getinterval sw
pop neg xs add /xs exch def} def
/accspe [ 65 /plusminus 66 /bar 67 /existential 68 /universal
69 /exclam 70 /numbersign 71 /greater 72 /question 73 /integral
74 /colon 75 /semicolon 76 /less 77 /bracketleft 78 /bracketright
79 /greaterequal 80 /braceleft 81 /braceright 82 /radical
83 /spade 84 /heart 85 /diamond 86 /club 87 /lessequal
88 /multiply 89 /percent 90 /infinity 48 /circlemultiply 49 /circleplus
50 /emptyset 51 /lozenge 52 /bullet 53 /arrowright 54 /arrowup
55 /arrowleft 56 /arrowdown 57 /arrowboth 48 /degree 44 /comma 43 /plus
 45 /angle 42 /angleleft 47 /divide 61 /notequal 40 /equivalence 41 /second
 97 /approxequal 98 /congruent 99 /perpendicular 100 /partialdiff 101 /florin
 102 /intersection 103 /union 104 /propersuperset 105 /reflexsuperset
 106 /notsubset 107 /propersubset 108 /reflexsubset 109 /element 110 /notelement
 111 /gradient 112 /logicaland 113 /logicalor 114 /arrowdblboth
 115 /arrowdblleft 116 /arrowdblup 117 /arrowdblright 118 /arrowdbldown
 119 /ampersand 120 /omega1 121 /similar 122 /aleph ] def
/Symbol /Special accspe ReEncode
gsave .25 .25 scale
%%EndProlog
 gsave 0 0 t 0 0 0 c [] 0 sd 1 lw 5 lw 1 284 m 1 284 l 2 284 l 2 283 l 3 283 l 4
 282 l 5 281 l 6 280 l 6 279 l 7 277 l 8 276 l 9 275 l 10 273 l 10 272 l 11 271
 l 12 269 l 13 268 l 14 267 l 14 265 l 15 264 l 16 263 l 17 263 l 17 262 l 18
 262 l 19 261 l 20 261 l 21 261 l 21 262 l 22 262 l 23 263 l 24 263 l 25 264 l
 25 265 l 26 267 l 27 268 l 28 269 l 29 271 l 29 272 l 30 273 l 31 275 l 32 276
 l 33 277 l 33 279 l 34 280 l 35 281 l 36 282 l 37 283 l 38 284 l 39 284 l 40
 284 l s 40 284 m 41 284 l 42 283 l 43 283 l 44 282 l 45 281 l 45 280 l 46 279 l
 47 277 l 48 276 l 49 275 l 49 273 l 50 272 l 51 271 l 52 269 l 52 268 l 53 267
 l 54 265 l 55 264 l 56 263 l 57 262 l 58 262 l 59 261 l 60 261 l 61 262 l 62
 262 l 63 263 l 64 263 l 64 264 l 65 265 l 66 267 l 67 268 l 68 269 l 68 271 l
 69 272 l 70 273 l 71 275 l 72 276 l 72 277 l 73 279 l 74 280 l 75 281 l 76 282
 l 76 283 l 77 283 l 78 284 l 79 284 l 80 284 l s 80 284 m 148 373 l s 1 lw 121
 337 m 98 326 l 116 312 l f 5 lw 193 432 m 91 299 l s 1 lw 135 356 m 158 367 l
 140 381 l f 5 lw 80 284 m 193 432 l s 80 284 m 193 432 l s 80 284 m 193 136 l s
 193 432 m 193 136 l s 197 432 m 197 432 l 200 431 l 204 430 l 207 429 l 210 427
 l 213 426 l 216 423 l 218 421 l 221 419 l 223 416 l 224 413 l 225 410 l 226 408
 l 227 405 l 227 402 l 227 399 l 227 397 l 226 394 l 225 392 l 224 391 l 222 389
 l 221 388 l 219 387 l 218 386 l 216 386 l 214 386 l 212 387 l 211 388 l 209 389
 l 208 391 l 207 392 l 206 394 l 205 397 l 205 399 l 205 402 l 205 405 l 205 408
 l 206 410 l 208 413 l 209 416 l 211 419 l 213 421 l 216 423 l 218 426 l 221 427
 l 225 429 l 228 430 l 231 431 l 235 432 l 239 432 l s 239 432 m 242 432 l 246
 431 l 249 430 l 252 429 l 256 427 l 259 426 l 261 423 l 264 421 l 266 419 l 268
 416 l 270 413 l 271 410 l 272 408 l 272 405 l 272 402 l 272 399 l 272 397 l 271
 394 l 270 392 l 269 391 l 268 389 l 266 388 l 265 387 l 263 386 l 261 386 l 260
 386 l 258 387 l 256 388 l 255 389 l 253 391 l 252 392 l 251 394 l 251 397 l 250
 399 l 250 402 l 250 405 l 251 408 l 252 410 l 253 413 l 255 416 l 257 419 l 259
 421 l 261 423 l 264 426 l 267 427 l 270 429 l 273 430 l 277 431 l 280 432 l 284
 432 l s 284 432 m 288 432 l 291 431 l 295 430 l 298 429 l 301 427 l 304 426 l
 307 423 l 309 421 l 311 419 l 313 416 l 315 413 l 316 410 l 317 408 l 318 405 l
 318 402 l 318 399 l 317 397 l 317 394 l 316 392 l 315 391 l 313 389 l 312 388 l
 310 387 l 308 386 l 307 386 l 305 386 l 303 387 l 302 388 l 300 389 l 299 391 l
 298 392 l 297 394 l 296 397 l 296 399 l 296 402 l 296 405 l 296 408 l 297 410 l
 298 413 l 300 416 l 302 419 l 304 421 l 307 423 l 309 426 l 312 427 l 316 429 l
 319 430 l 322 431 l 326 432 l 329 432 l s 329 432 m 333 432 l 337 431 l 340 430
 l 343 429 l 347 427 l 350 426 l 352 423 l 355 421 l 357 419 l 359 416 l 360 413
 l 362 410 l 363 408 l 363 405 l 363 402 l 363 399 l 363 397 l 362 394 l 361 392
 l 360 391 l 359 389 l 357 388 l 356 387 l 354 386 l 352 386 l 350 386 l 349 387
 l 347 388 l 346 389 l 344 391 l 343 392 l 342 394 l 341 397 l 341 399 l 341 402
 l 341 405 l 342 408 l 343 410 l 344 413 l 345 416 l 347 419 l 350 421 l 352 423
 l 355 426 l 358 427 l 361 429 l 364 430 l 368 431 l 371 432 l 375 432 l s 371
 136 m 371 136 l 368 137 l 364 138 l 361 139 l 358 141 l 355 142 l 352 145 l 350
 147 l 347 149 l 345 152 l 344 155 l 343 158 l 342 160 l 341 163 l 341 166 l 341
 169 l 341 171 l 342 174 l 343 176 l 344 177 l 346 179 l 347 180 l 349 181 l 350
 182 l 352 182 l 354 182 l 356 181 l 357 180 l 359 179 l 360 177 l 361 176 l 362
 174 l 363 171 l 363 169 l 363 166 l 363 163 l 363 160 l 362 158 l 360 155 l 359
 152 l 357 149 l 355 147 l 352 145 l 350 142 l 347 141 l 343 139 l 340 138 l 337
 137 l 333 136 l 329 136 l s 329 136 m 326 136 l 322 137 l 319 138 l 316 139 l
 312 141 l 309 142 l 307 145 l 304 147 l 302 149 l 300 152 l 298 155 l 297 158 l
 296 160 l 296 163 l 296 166 l 296 169 l 296 171 l 297 174 l 298 176 l 299 177 l
 300 179 l 302 180 l 303 181 l 305 182 l 307 182 l 308 182 l 310 181 l 312 180 l
 313 179 l 315 177 l 316 176 l 317 174 l 317 171 l 318 169 l 318 166 l 318 163 l
 317 160 l 316 158 l 315 155 l 313 152 l 311 149 l 309 147 l 307 145 l 304 142 l
 301 141 l 298 139 l 295 138 l 291 137 l 288 136 l 284 136 l s 284 136 m 280 136
 l 277 137 l 273 138 l 270 139 l 267 141 l 264 142 l 261 145 l 259 147 l 257 149
 l 255 152 l 253 155 l 252 158 l 251 160 l 250 163 l 250 166 l 250 169 l 251 171
 l 251 174 l 252 176 l 253 177 l 255 179 l 256 180 l 258 181 l 260 182 l 261 182
 l 263 182 l 265 181 l 266 180 l 268 179 l 269 177 l 270 176 l 271 174 l 272 171
 l 272 169 l 272 166 l 272 163 l 272 160 l 271 158 l 270 155 l 268 152 l 266 149
 l 264 147 l 261 145 l 259 142 l 256 141 l 252 139 l 249 138 l 246 137 l 242 136
 l 239 136 l s 239 136 m 235 136 l 231 137 l 228 138 l 225 139 l 221 141 l 218
 142 l 216 145 l 213 147 l 211 149 l 209 152 l 208 155 l 206 158 l 205 160 l 205
 163 l 205 166 l 205 169 l 205 171 l 206 174 l 207 176 l 208 177 l 209 179 l 211
 180 l 212 181 l 214 182 l 216 182 l 218 182 l 219 181 l 221 180 l 222 179 l 224
 177 l 225 176 l 226 174 l 227 171 l 227 169 l 227 166 l 227 163 l 226 160 l 225
 158 l 224 155 l 223 152 l 221 149 l 218 147 l 216 145 l 213 142 l 210 141 l 207
 139 l 204 138 l 200 137 l 197 136 l 193 136 l s 375 432 m 375 136 l s 375 432 m
 488 284 l s 375 136 m 488 284 l s 489 284 m 489 284 l 490 284 l 491 283 l 492
 283 l 492 282 l 493 281 l 494 280 l 495 279 l 496 277 l 496 276 l 497 275 l 498
 273 l 499 272 l 500 271 l 500 269 l 501 268 l 502 267 l 503 265 l 504 264 l 504
 263 l 505 263 l 506 262 l 507 262 l 508 261 l 509 261 l 510 262 l 511 262 l 512
 263 l 513 264 l 514 265 l 515 267 l 516 268 l 516 269 l 517 271 l 518 272 l 519
 273 l 519 275 l 520 276 l 521 277 l 522 279 l 523 280 l 523 281 l 524 282 l 525
 283 l 526 283 l 527 284 l 528 284 l s 528 284 m 529 284 l 530 284 l 531 283 l
 532 282 l 533 281 l 534 280 l 535 279 l 535 277 l 536 276 l 537 275 l 538 273 l
 539 272 l 539 271 l 540 269 l 541 268 l 542 267 l 543 265 l 543 264 l 544 263 l
 545 263 l 546 262 l 547 262 l 547 261 l 548 261 l 549 261 l 550 262 l 551 262 l
 551 263 l 552 263 l 553 264 l 554 265 l 554 267 l 555 268 l 556 269 l 557 271 l
 558 272 l 558 273 l 559 275 l 560 276 l 561 277 l 562 279 l 562 280 l 563 281 l
 564 282 l 565 283 l 566 283 l 566 284 l 567 284 l 568 284 l s 1 lw 273 84 m 271
 82 l 269 78 l 267 74 l 266 69 l 266 64 l 267 59 l 269 55 l 271 51 l 273 49 l s
 293 72 m 293 57 l s 293 69 m 290 71 l 288 72 l 285 72 l 283 71 l 281 69 l 280
 65 l 280 63 l 281 60 l 283 58 l 285 57 l 288 57 l 290 58 l 293 60 l s 300 84 m
 302 82 l 305 78 l 307 74 l 308 69 l 308 64 l 307 59 l 305 55 l 302 51 l 300 49
 l s
gr gr
end
%%EOF
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% gh.eps %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%!PS-Adobe-2.0 EPSF-2.0
%%BoundingBox: 0 0 142 142
%%Title: /usr/users/ak/PAW/SINGLET/gh.eps
%%Creator: HIGZ Version 1.16/11
%%CreationDate: 93/10/03   12.44
%%EndComments
80 dict begin
/s {stroke} def /l {lineto} def /m {moveto} def /t {translate} def
/sw {stringwidth} def /r {rotate} def /rl {roll} def
/d {rlineto} def /rm {rmoveto} def /gr {grestore} def /f {eofill} def
/c {setrgbcolor} def /lw {setlinewidth} def /sd {setdash} def
/cl {closepath} def /sf {scalefont setfont} def
/box {m dup 0 exch d exch 0 d 0 exch neg d cl} def
/bl {box s} def /bf {box f} def
/mp {newpath /y exch def /x exch def} def
/side {[w .77 mul w .23 mul] .385 w mul sd w 0 l currentpoint t -144 r} def
/mr {mp x y w2 0 360 arc} def /m24 {mr s} def /m20 {mr f} def
/mb {mp x y w2 add m w2 neg 0 d 0 w neg d w 0 d 0 w d cl} def
/mt {mp x y w2 add m w2 neg w neg d w 0 d cl} def
/m21 {mb f} def /m25 {mb s} def /m22 {mt f} def /m26{mt s} def
/m23 {mp x y w2 sub m w2 w d w neg 0 d cl f} def
 /m27 {mp x y w2 add m w3 neg w2 neg d w3 w2 neg d w3 w2 d cl s} def
 /m28 {mp x w2 sub y w2 sub w3 add m w3 0 d 0 w3 neg d w3 0 d 0 w3 d w3 0 d
 0 w3 d w3 neg 0 d 0 w3 d w3 neg 0 d 0 w3 neg d w3 neg 0 d cl s } def
 /m29 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 4 {side} repeat cl fill gr} def
 /m30 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 5 {side} repeat s gr} def /m31 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d
 x w2 sub y w2 add m w w neg d x w2 sub y w2
 sub m w w d s} def
/m2 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d s} def
/m5 {mp x w2 sub y w2 sub m w w d x w2 sub y w2 add m w w neg d s} def
/reencdict 24 dict def /ReEncode {reencdict begin /nco&na exch def
/nfnam exch def /basefontname exch def /basefontdict basefontname findfont def
/newfont basefontdict maxlength dict def basefontdict {exch dup /FID ne
{dup /Encoding eq {exch dup length array copy newfont 3 1 roll put} {exch
newfont 3 1 roll put} ifelse} {pop pop} ifelse } forall newfont
/FontName nfnam put nco&na aload pop nco&na length 2 idiv {newfont
/Encoding get 3 1 roll put} repeat nfnam newfont definefont pop end } def
/accvec [ 176 /agrave 181 /Agrave 190 /acircumflex 192 /Acircumflex
201 /adieresis 204 /Adieresis 209 /ccedilla 210 /Ccedilla 211 /eacute
212 /Eacute 213 /egrave 214 /Egrave 215 /ecircumflex 216 /Ecircumflex
217 /edieresis 218 /Edieresis 219 /icircumflex 220 /Icircumflex
221 /idieresis 222 /Idieresis 223 /ntilde 224 /Ntilde 226 /ocircumflex
228 /Ocircumflex 229 /odieresis 230 /Odieresis 231 /ucircumflex 236 /Ucircumflex
237 /udieresis 238 /Udieresis 239 /aring 242 /Aring 243 /ydieresis
244 /Ydieresis 246 /aacute 247 /Aacute 252 /ugrave 253 /Ugrave] def
/Times-Roman /Times-Roman accvec ReEncode
/Times-Italic /Times-Italic accvec ReEncode
/Times-Bold /Times-Bold accvec ReEncode
/Times-BoldItalic /Times-BoldItalic accvec ReEncode
/Helvetica /Helvetica accvec ReEncode
/Helvetica-Oblique /Helvetica-Oblique accvec ReEncode
/Helvetica-Bold /Helvetica-Bold accvec ReEncode
/Helvetica-BoldOblique /Helvetica-BoldOblique  accvec ReEncode
/Courier /Courier accvec ReEncode
/Courier-Oblique /Courier-Oblique accvec ReEncode
/Courier-Bold /Courier-Bold accvec ReEncode
/Courier-BoldOblique /Courier-BoldOblique accvec ReEncode
/oshow {gsave [] 0 sd true charpath stroke gr} def
/stwn { /fs exch def /fn exch def /text exch def fn findfont fs sf
 text sw pop xs add /xs exch def} def
/stwb { /fs exch def /fn exch def /nbas exch def /textf exch def
textf length /tlen exch def nbas tlen gt {/nbas tlendef} if
fn findfont fs sf textf dup length nbas sub nbas getinterval sw
pop neg xs add /xs exch def} def
/accspe [ 65 /plusminus 66 /bar 67 /existential 68 /universal
69 /exclam 70 /numbersign 71 /greater 72 /question 73 /integral
74 /colon 75 /semicolon 76 /less 77 /bracketleft 78 /bracketright
79 /greaterequal 80 /braceleft 81 /braceright 82 /radical
83 /spade 84 /heart 85 /diamond 86 /club 87 /lessequal
88 /multiply 89 /percent 90 /infinity 48 /circlemultiply 49 /circleplus
50 /emptyset 51 /lozenge 52 /bullet 53 /arrowright 54 /arrowup
55 /arrowleft 56 /arrowdown 57 /arrowboth 48 /degree 44 /comma 43 /plus
 45 /angle 42 /angleleft 47 /divide 61 /notequal 40 /equivalence 41 /second
 97 /approxequal 98 /congruent 99 /perpendicular 100 /partialdiff 101 /florin
 102 /intersection 103 /union 104 /propersuperset 105 /reflexsuperset
 106 /notsubset 107 /propersubset 108 /reflexsubset 109 /element 110 /notelement
 111 /gradient 112 /logicaland 113 /logicalor 114 /arrowdblboth
 115 /arrowdblleft 116 /arrowdblup 117 /arrowdblright 118 /arrowdbldown
 119 /ampersand 120 /omega1 121 /similar 122 /aleph ] def
/Symbol /Special accspe ReEncode
gsave .25 .25 scale
%%EndProlog
 gsave 0 0 t 0 0 0 c [] 0 sd 1 lw 5 lw 1 284 m 1 284 l 2 284 l 2 283 l 3 283 l 4
 282 l 5 281 l 6 280 l 6 279 l 7 277 l 8 276 l 9 275 l 10 273 l 10 272 l 11 271
 l 12 269 l 13 268 l 14 267 l 14 265 l 15 264 l 16 263 l 17 263 l 17 262 l 18
 262 l 19 261 l 20 261 l 21 261 l 21 262 l 22 262 l 23 263 l 24 263 l 25 264 l
 25 265 l 26 267 l 27 268 l 28 269 l 29 271 l 29 272 l 30 273 l 31 275 l 32 276
 l 33 277 l 33 279 l 34 280 l 35 281 l 36 282 l 37 283 l 38 284 l 39 284 l 40
 284 l s 40 284 m 41 284 l 42 283 l 43 283 l 44 282 l 45 281 l 45 280 l 46 279 l
 47 277 l 48 276 l 49 275 l 49 273 l 50 272 l 51 271 l 52 269 l 52 268 l 53 267
 l 54 265 l 55 264 l 56 263 l 57 262 l 58 262 l 59 261 l 60 261 l 61 262 l 62
 262 l 63 263 l 64 263 l 64 264 l 65 265 l 66 267 l 67 268 l 68 269 l 68 271 l
 69 272 l 70 273 l 71 275 l 72 276 l 72 277 l 73 279 l 74 280 l 75 281 l 76 282
 l 76 283 l 77 283 l 78 284 l 79 284 l 80 284 l s 20 lw 80 284 m 193 432 l s 80
 284 m 193 136 l s 193 432 m 193 136 l s 5 lw 197 432 m 197 432 l 200 431 l 204
 430 l 207 429 l 210 427 l 213 426 l 216 423 l 218 421 l 221 419 l 223 416 l 224
 413 l 225 410 l 226 408 l 227 405 l 227 402 l 227 399 l 227 397 l 226 394 l 225
 392 l 224 391 l 222 389 l 221 388 l 219 387 l 218 386 l 216 386 l 214 386 l 212
 387 l 211 388 l 209 389 l 208 391 l 207 392 l 206 394 l 205 397 l 205 399 l 205
 402 l 205 405 l 205 408 l 206 410 l 208 413 l 209 416 l 211 419 l 213 421 l 216
 423 l 218 426 l 221 427 l 225 429 l 228 430 l 231 431 l 235 432 l 239 432 l s
 239 432 m 242 432 l 246 431 l 249 430 l 252 429 l 256 427 l 259 426 l 261 423 l
 264 421 l 266 419 l 268 416 l 270 413 l 271 410 l 272 408 l 272 405 l 272 402 l
 272 399 l 272 397 l 271 394 l 270 392 l 269 391 l 268 389 l 266 388 l 265 387 l
 263 386 l 261 386 l 260 386 l 258 387 l 256 388 l 255 389 l 253 391 l 252 392 l
 251 394 l 251 397 l 250 399 l 250 402 l 250 405 l 251 408 l 252 410 l 253 413 l
 255 416 l 257 419 l 259 421 l 261 423 l 264 426 l 267 427 l 270 429 l 273 430 l
 277 431 l 280 432 l 284 432 l s 284 432 m 288 432 l 291 431 l 295 430 l 298 429
 l 301 427 l 304 426 l 307 423 l 309 421 l 311 419 l 313 416 l 315 413 l 316 410
 l 317 408 l 318 405 l 318 402 l 318 399 l 317 397 l 317 394 l 316 392 l 315 391
 l 313 389 l 312 388 l 310 387 l 308 386 l 307 386 l 305 386 l 303 387 l 302 388
 l 300 389 l 299 391 l 298 392 l 297 394 l 296 397 l 296 399 l 296 402 l 296 405
 l 296 408 l 297 410 l 298 413 l 300 416 l 302 419 l 304 421 l 307 423 l 309 426
 l 312 427 l 316 429 l 319 430 l 322 431 l 326 432 l 329 432 l s 329 432 m 333
 432 l 337 431 l 340 430 l 343 429 l 347 427 l 350 426 l 352 423 l 355 421 l 357
 419 l 359 416 l 360 413 l 362 410 l 363 408 l 363 405 l 363 402 l 363 399 l 363
 397 l 362 394 l 361 392 l 360 391 l 359 389 l 357 388 l 356 387 l 354 386 l 352
 386 l 350 386 l 349 387 l 347 388 l 346 389 l 344 391 l 343 392 l 342 394 l 341
 397 l 341 399 l 341 402 l 341 405 l 342 408 l 343 410 l 344 413 l 345 416 l 347
 419 l 350 421 l 352 423 l 355 426 l 358 427 l 361 429 l 364 430 l 368 431 l 371
 432 l 375 432 l s 371 136 m 371 136 l 368 137 l 364 138 l 361 139 l 358 141 l
 355 142 l 352 145 l 350 147 l 347 149 l 345 152 l 344 155 l 343 158 l 342 160 l
 341 163 l 341 166 l 341 169 l 341 171 l 342 174 l 343 176 l 344 177 l 346 179 l
 347 180 l 349 181 l 350 182 l 352 182 l 354 182 l 356 181 l 357 180 l 359 179 l
 360 177 l 361 176 l 362 174 l 363 171 l 363 169 l 363 166 l 363 163 l 363 160 l
 362 158 l 360 155 l 359 152 l 357 149 l 355 147 l 352 145 l 350 142 l 347 141 l
 343 139 l 340 138 l 337 137 l 333 136 l 329 136 l s 329 136 m 326 136 l 322 137
 l 319 138 l 316 139 l 312 141 l 309 142 l 307 145 l 304 147 l 302 149 l 300 152
 l 298 155 l 297 158 l 296 160 l 296 163 l 296 166 l 296 169 l 296 171 l 297 174
 l 298 176 l 299 177 l 300 179 l 302 180 l 303 181 l 305 182 l 307 182 l 308 182
 l 310 181 l 312 180 l 313 179 l 315 177 l 316 176 l 317 174 l 317 171 l 318 169
 l 318 166 l 318 163 l 317 160 l 316 158 l 315 155 l 313 152 l 311 149 l 309 147
 l 307 145 l 304 142 l 301 141 l 298 139 l 295 138 l 291 137 l 288 136 l 284 136
 l s 284 136 m 280 136 l 277 137 l 273 138 l 270 139 l 267 141 l 264 142 l 261
 145 l 259 147 l 257 149 l 255 152 l 253 155 l 252 158 l 251 160 l 250 163 l 250
 166 l 250 169 l 251 171 l 251 174 l 252 176 l 253 177 l 255 179 l 256 180 l 258
 181 l 260 182 l 261 182 l 263 182 l 265 181 l 266 180 l 268 179 l 269 177 l 270
 176 l 271 174 l 272 171 l 272 169 l 272 166 l 272 163 l 272 160 l 271 158 l 270
 155 l 268 152 l 266 149 l 264 147 l 261 145 l 259 142 l 256 141 l 252 139 l 249
 138 l 246 137 l 242 136 l 239 136 l s 239 136 m 235 136 l 231 137 l 228 138 l
 225 139 l 221 141 l 218 142 l 216 145 l 213 147 l 211 149 l 209 152 l 208 155 l
 206 158 l 205 160 l 205 163 l 205 166 l 205 169 l 205 171 l 206 174 l 207 176 l
 208 177 l 209 179 l 211 180 l 212 181 l 214 182 l 216 182 l 218 182 l 219 181 l
 221 180 l 222 179 l 224 177 l 225 176 l 226 174 l 227 171 l 227 169 l 227 166 l
 227 163 l 226 160 l 225 158 l 224 155 l 223 152 l 221 149 l 218 147 l 216 145 l
 213 142 l 210 141 l 207 139 l 204 138 l 200 137 l 197 136 l 193 136 l s 375 432
 m 375 136 l s 375 432 m 488 284 l s 375 136 m 488 284 l s 489 284 m 489 284 l
 490 284 l 491 283 l 492 283 l 492 282 l 493 281 l 494 280 l 495 279 l 496 277 l
 496 276 l 497 275 l 498 273 l 499 272 l 500 271 l 500 269 l 501 268 l 502 267 l
 503 265 l 504 264 l 504 263 l 505 263 l 506 262 l 507 262 l 508 261 l 509 261 l
 510 262 l 511 262 l 512 263 l 513 264 l 514 265 l 515 267 l 516 268 l 516 269 l
 517 271 l 518 272 l 519 273 l 519 275 l 520 276 l 521 277 l 522 279 l 523 280 l
 523 281 l 524 282 l 525 283 l 526 283 l 527 284 l 528 284 l s 528 284 m 529 284
 l 530 284 l 531 283 l 532 282 l 533 281 l 534 280 l 535 279 l 535 277 l 536 276
 l 537 275 l 538 273 l 539 272 l 539 271 l 540 269 l 541 268 l 542 267 l 543 265
 l 543 264 l 544 263 l 545 263 l 546 262 l 547 262 l 547 261 l 548 261 l 549 261
 l 550 262 l 551 262 l 551 263 l 552 263 l 553 264 l 554 265 l 554 267 l 555 268
 l 556 269 l 557 271 l 558 272 l 558 273 l 559 275 l 560 276 l 561 277 l 562 279
 l 562 280 l 563 281 l 564 282 l 565 283 l 566 283 l 566 284 l 567 284 l 568 284
 l s 1 lw 270 102 m 270 57 l s 270 102 m 296 102 l s 302 68 m 302 45 l s 302 56
 m 306 60 l 308 61 l 311 61 l 313 60 l 314 56 l 314 45 l s
gr gr
end
%%EOF
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% gh1.eps
%!PS-Adobe-2.0 EPSF-2.0
%%BoundingBox: 0 0 142 142
%%Title: /usr/users/ak/PAW/SINGLET/gh1.eps
%%Creator: HIGZ Version 1.16/11
%%CreationDate: 93/10/03   12.45
%%EndComments
80 dict begin
/s {stroke} def /l {lineto} def /m {moveto} def /t {translate} def
/sw {stringwidth} def /r {rotate} def /rl {roll} def
/d {rlineto} def /rm {rmoveto} def /gr {grestore} def /f {eofill} def
/c {setrgbcolor} def /lw {setlinewidth} def /sd {setdash} def
/cl {closepath} def /sf {scalefont setfont} def
/box {m dup 0 exch d exch 0 d 0 exch neg d cl} def
/bl {box s} def /bf {box f} def
/mp {newpath /y exch def /x exch def} def
/side {[w .77 mul w .23 mul] .385 w mul sd w 0 l currentpoint t -144 r} def
/mr {mp x y w2 0 360 arc} def /m24 {mr s} def /m20 {mr f} def
/mb {mp x y w2 add m w2 neg 0 d 0 w neg d w 0 d 0 w d cl} def
/mt {mp x y w2 add m w2 neg w neg d w 0 d cl} def
/m21 {mb f} def /m25 {mb s} def /m22 {mt f} def /m26{mt s} def
/m23 {mp x y w2 sub m w2 w d w neg 0 d cl f} def
 /m27 {mp x y w2 add m w3 neg w2 neg d w3 w2 neg d w3 w2 d cl s} def
 /m28 {mp x w2 sub y w2 sub w3 add m w3 0 d 0 w3 neg d w3 0 d 0 w3 d w3 0 d
 0 w3 d w3 neg 0 d 0 w3 d w3 neg 0 d 0 w3 neg d w3 neg 0 d cl s } def
 /m29 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 4 {side} repeat cl fill gr} def
 /m30 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 5 {side} repeat s gr} def /m31 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d
 x w2 sub y w2 add m w w neg d x w2 sub y w2
 sub m w w d s} def
/m2 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d s} def
/m5 {mp x w2 sub y w2 sub m w w d x w2 sub y w2 add m w w neg d s} def
/reencdict 24 dict def /ReEncode {reencdict begin /nco&na exch def
/nfnam exch def /basefontname exch def /basefontdict basefontname findfont def
/newfont basefontdict maxlength dict def basefontdict {exch dup /FID ne
{dup /Encoding eq {exch dup length array copy newfont 3 1 roll put} {exch
newfont 3 1 roll put} ifelse} {pop pop} ifelse } forall newfont
/FontName nfnam put nco&na aload pop nco&na length 2 idiv {newfont
/Encoding get 3 1 roll put} repeat nfnam newfont definefont pop end } def
/accvec [ 176 /agrave 181 /Agrave 190 /acircumflex 192 /Acircumflex
201 /adieresis 204 /Adieresis 209 /ccedilla 210 /Ccedilla 211 /eacute
212 /Eacute 213 /egrave 214 /Egrave 215 /ecircumflex 216 /Ecircumflex
217 /edieresis 218 /Edieresis 219 /icircumflex 220 /Icircumflex
221 /idieresis 222 /Idieresis 223 /ntilde 224 /Ntilde 226 /ocircumflex
228 /Ocircumflex 229 /odieresis 230 /Odieresis 231 /ucircumflex 236 /Ucircumflex
237 /udieresis 238 /Udieresis 239 /aring 242 /Aring 243 /ydieresis
244 /Ydieresis 246 /aacute 247 /Aacute 252 /ugrave 253 /Ugrave] def
/Times-Roman /Times-Roman accvec ReEncode
/Times-Italic /Times-Italic accvec ReEncode
/Times-Bold /Times-Bold accvec ReEncode
/Times-BoldItalic /Times-BoldItalic accvec ReEncode
/Helvetica /Helvetica accvec ReEncode
/Helvetica-Oblique /Helvetica-Oblique accvec ReEncode
/Helvetica-Bold /Helvetica-Bold accvec ReEncode
/Helvetica-BoldOblique /Helvetica-BoldOblique  accvec ReEncode
/Courier /Courier accvec ReEncode
/Courier-Oblique /Courier-Oblique accvec ReEncode
/Courier-Bold /Courier-Bold accvec ReEncode
/Courier-BoldOblique /Courier-BoldOblique accvec ReEncode
/oshow {gsave [] 0 sd true charpath stroke gr} def
/stwn { /fs exch def /fn exch def /text exch def fn findfont fs sf
 text sw pop xs add /xs exch def} def
/stwb { /fs exch def /fn exch def /nbas exch def /textf exch def
textf length /tlen exch def nbas tlen gt {/nbas tlendef} if
fn findfont fs sf textf dup length nbas sub nbas getinterval sw
pop neg xs add /xs exch def} def
/accspe [ 65 /plusminus 66 /bar 67 /existential 68 /universal
69 /exclam 70 /numbersign 71 /greater 72 /question 73 /integral
74 /colon 75 /semicolon 76 /less 77 /bracketleft 78 /bracketright
79 /greaterequal 80 /braceleft 81 /braceright 82 /radical
83 /spade 84 /heart 85 /diamond 86 /club 87 /lessequal
88 /multiply 89 /percent 90 /infinity 48 /circlemultiply 49 /circleplus
50 /emptyset 51 /lozenge 52 /bullet 53 /arrowright 54 /arrowup
55 /arrowleft 56 /arrowdown 57 /arrowboth 48 /degree 44 /comma 43 /plus
 45 /angle 42 /angleleft 47 /divide 61 /notequal 40 /equivalence 41 /second
 97 /approxequal 98 /congruent 99 /perpendicular 100 /partialdiff 101 /florin
 102 /intersection 103 /union 104 /propersuperset 105 /reflexsuperset
 106 /notsubset 107 /propersubset 108 /reflexsubset 109 /element 110 /notelement
 111 /gradient 112 /logicaland 113 /logicalor 114 /arrowdblboth
 115 /arrowdblleft 116 /arrowdblup 117 /arrowdblright 118 /arrowdbldown
 119 /ampersand 120 /omega1 121 /similar 122 /aleph ] def
/Symbol /Special accspe ReEncode
gsave .25 .25 scale
%%EndProlog
 gsave 0 0 t 0 0 0 c [] 0 sd 1 lw 5 lw 1 284 m 1 284 l 2 284 l 2 283 l 3 283 l 4
 282 l 5 281 l 5 280 l 6 279 l 7 277 l 8 276 l 8 275 l 9 273 l 10 272 l 11 271 l
 11 269 l 12 268 l 13 267 l 14 265 l 14 264 l 15 263 l 16 263 l 17 262 l 18 261
 l 19 261 l 20 261 l 20 262 l 21 262 l 22 263 l 23 263 l 23 264 l 24 265 l 25
 267 l 26 268 l 27 269 l 27 271 l 28 272 l 29 273 l 30 275 l 30 276 l 31 277 l
 32 279 l 33 280 l 33 281 l 34 282 l 35 283 l 36 283 l 36 284 l 37 284 l 38 284
 l s 38 284 m 39 284 l 40 283 l 41 283 l 42 282 l 42 281 l 43 280 l 44 279 l 45
 277 l 45 276 l 46 275 l 47 273 l 48 272 l 48 271 l 49 269 l 50 268 l 51 267 l
 51 265 l 52 264 l 53 263 l 54 263 l 55 262 l 56 261 l 57 261 l 58 261 l 58 262
 l 59 262 l 60 263 l 61 263 l 61 264 l 62 265 l 63 267 l 64 268 l 64 269 l 65
 271 l 66 272 l 67 273 l 67 275 l 68 276 l 69 277 l 70 279 l 70 280 l 71 281 l
 72 282 l 73 283 l 74 284 l 75 284 l 76 284 l s 76 284 m 76 284 l 77 284 l 78
 283 l 79 283 l 80 282 l 80 281 l 81 280 l 82 279 l 83 277 l 83 276 l 84 275 l
 85 273 l 86 272 l 86 271 l 87 269 l 88 268 l 89 267 l 89 265 l 90 264 l 91 263
 l 92 263 l 92 262 l 93 262 l 94 261 l 95 261 l 96 262 l 97 262 l 98 263 l 99
 264 l 100 265 l 101 267 l 101 268 l 102 269 l 103 271 l 104 272 l 105 273 l 105
 275 l 106 276 l 107 277 l 108 279 l 108 280 l 109 281 l 110 282 l 111 283 l 112
 284 l 113 284 l 114 284 l s 1 lw 114 307 m 111 307 l 108 306 l 105 305 l 103
 304 l 100 302 l 98 301 l 96 298 l 94 296 l 93 294 l 92 291 l 91 288 l 91 285 l
 91 283 l 91 280 l 92 277 l 93 274 l 94 272 l 96 270 l 98 267 l 100 266 l 103
 264 l 105 263 l 108 262 l 111 261 l 114 261 l 116 261 l 119 262 l 122 263 l 125
 264 l 127 266 l 129 267 l 131 270 l 133 272 l 134 274 l 135 277 l 136 280 l 136
 283 l 136 285 l 136 288 l 135 291 l 134 294 l 133 296 l 131 298 l 129 301 l 127
 302 l 125 304 l 122 305 l 119 306 l 116 307 l 114 307 l f 5 lw 454 284 m 454
 295 l 453 306 l 451 316 l 449 327 l 446 337 l 442 347 l 438 357 l 433 367 l 427
 376 l 421 385 l 415 394 l 407 402 l 400 409 l 391 416 l 383 423 l 374 429 l 364
 434 l 355 439 l 345 443 l 335 447 l 324 450 l 314 452 l 303 453 l 292 454 l 281
 454 l 270 454 l 260 453 l 249 451 l 239 448 l 228 445 l 218 441 l 208 437 l 199
 432 l 190 426 l 181 420 l 172 413 l 164 405 l 157 398 l 150 389 l 144 381 l 138
 372 l 133 362 l 128 352 l 124 342 l 121 332 l 118 322 l 116 311 l 114 300 l 114
 289 l 114 279 l 114 268 l 116 257 l 118 246 l 121 236 l 124 226 l 128 216 l 133
 206 l 138 196 l 144 187 l 150 179 l 157 170 l 164 163 l 172 155 l 181 148 l 190
 142 l 199 136 l 208 131 l 218 127 l 228 123 l 239 120 l 249 117 l 260 115 l 270
 114 l 281 114 l 292 114 l 303 115 l 314 116 l 324 118 l 335 121 l 345 125 l 355
 129 l 364 134 l 374 139 l 383 145 l 391 152 l 400 159 l 407 166 l 415 174 l 421
 183 l 427 192 l 433 201 l 438 211 l 442 221 l 446 231 l 449 241 l 451 252 l 453
 262 l 454 273 l 454 284 l s 455 284 m 455 284 l 456 284 l 457 283 l 458 282 l
 459 281 l 460 280 l 460 279 l 461 277 l 462 276 l 463 275 l 463 273 l 464 272 l
 465 271 l 466 269 l 467 268 l 467 267 l 468 265 l 469 264 l 470 263 l 471 262 l
 472 262 l 473 261 l 474 261 l 475 262 l 476 262 l 476 263 l 477 263 l 478 264 l
 479 265 l 479 267 l 480 268 l 481 269 l 482 271 l 482 272 l 483 273 l 484 275 l
 485 276 l 485 277 l 486 279 l 487 280 l 488 281 l 488 282 l 489 283 l 490 283 l
 491 284 l 492 284 l s 492 284 m 493 284 l 494 284 l 495 283 l 496 282 l 497 281
 l 498 280 l 498 279 l 499 277 l 500 276 l 501 275 l 501 273 l 502 272 l 503 271
 l 504 269 l 504 268 l 505 267 l 506 265 l 507 264 l 507 263 l 508 263 l 509 262
 l 510 262 l 510 261 l 511 261 l 512 261 l 513 262 l 514 263 l 515 263 l 516 264
 l 517 265 l 517 267 l 518 268 l 519 269 l 520 271 l 520 272 l 521 273 l 522 275
 l 523 276 l 523 277 l 524 279 l 525 280 l 526 281 l 526 282 l 527 283 l 528 283
 l 529 284 l 530 284 l s 530 284 m 531 284 l 532 284 l 532 283 l 533 283 l 534
 282 l 535 281 l 535 280 l 536 279 l 537 277 l 538 276 l 538 275 l 539 273 l 540
 272 l 541 271 l 541 269 l 542 268 l 543 267 l 544 265 l 545 264 l 545 263 l 546
 263 l 547 262 l 548 262 l 548 261 l 549 261 l 550 261 l 551 262 l 552 263 l 553
 263 l 554 264 l 554 265 l 555 267 l 556 268 l 557 269 l 557 271 l 558 272 l 559
 273 l 560 275 l 560 276 l 561 277 l 562 279 l 563 280 l 563 281 l 564 282 l 565
 283 l 566 283 l 566 284 l 567 284 l 568 284 l s 1 lw 281 102 m 281 57 l s 281
 102 m 307 102 l s 314 68 m 314 45 l s 314 56 m 317 60 l 319 61 l 322 61 l 325
 60 l 326 56 l 326 45 l s 336 64 m 339 65 l 342 68 l 342 45 l s
gr gr
end
%%EOF
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% gh2.eps %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%!PS-Adobe-2.0 EPSF-2.0
%%BoundingBox: 0 0 142 142
%%Title: /usr/users/ak/PAW/SINGLET/gh2.eps
%%Creator: HIGZ Version 1.16/11
%%CreationDate: 93/10/03   12.45
%%EndComments
80 dict begin
/s {stroke} def /l {lineto} def /m {moveto} def /t {translate} def
/sw {stringwidth} def /r {rotate} def /rl {roll} def
/d {rlineto} def /rm {rmoveto} def /gr {grestore} def /f {eofill} def
/c {setrgbcolor} def /lw {setlinewidth} def /sd {setdash} def
/cl {closepath} def /sf {scalefont setfont} def
/box {m dup 0 exch d exch 0 d 0 exch neg d cl} def
/bl {box s} def /bf {box f} def
/mp {newpath /y exch def /x exch def} def
/side {[w .77 mul w .23 mul] .385 w mul sd w 0 l currentpoint t -144 r} def
/mr {mp x y w2 0 360 arc} def /m24 {mr s} def /m20 {mr f} def
/mb {mp x y w2 add m w2 neg 0 d 0 w neg d w 0 d 0 w d cl} def
/mt {mp x y w2 add m w2 neg w neg d w 0 d cl} def
/m21 {mb f} def /m25 {mb s} def /m22 {mt f} def /m26{mt s} def
/m23 {mp x y w2 sub m w2 w d w neg 0 d cl f} def
 /m27 {mp x y w2 add m w3 neg w2 neg d w3 w2 neg d w3 w2 d cl s} def
 /m28 {mp x w2 sub y w2 sub w3 add m w3 0 d 0 w3 neg d w3 0 d 0 w3 d w3 0 d
 0 w3 d w3 neg 0 d 0 w3 d w3 neg 0 d 0 w3 neg d w3 neg 0 d cl s } def
 /m29 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 4 {side} repeat cl fill gr} def
 /m30 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 5 {side} repeat s gr} def /m31 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d
 x w2 sub y w2 add m w w neg d x w2 sub y w2
 sub m w w d s} def
/m2 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d s} def
/m5 {mp x w2 sub y w2 sub m w w d x w2 sub y w2 add m w w neg d s} def
/reencdict 24 dict def /ReEncode {reencdict begin /nco&na exch def
/nfnam exch def /basefontname exch def /basefontdict basefontname findfont def
/newfont basefontdict maxlength dict def basefontdict {exch dup /FID ne
{dup /Encoding eq {exch dup length array copy newfont 3 1 roll put} {exch
newfont 3 1 roll put} ifelse} {pop pop} ifelse } forall newfont
/FontName nfnam put nco&na aload pop nco&na length 2 idiv {newfont
/Encoding get 3 1 roll put} repeat nfnam newfont definefont pop end } def
/accvec [ 176 /agrave 181 /Agrave 190 /acircumflex 192 /Acircumflex
201 /adieresis 204 /Adieresis 209 /ccedilla 210 /Ccedilla 211 /eacute
212 /Eacute 213 /egrave 214 /Egrave 215 /ecircumflex 216 /Ecircumflex
217 /edieresis 218 /Edieresis 219 /icircumflex 220 /Icircumflex
221 /idieresis 222 /Idieresis 223 /ntilde 224 /Ntilde 226 /ocircumflex
228 /Ocircumflex 229 /odieresis 230 /Odieresis 231 /ucircumflex 236 /Ucircumflex
237 /udieresis 238 /Udieresis 239 /aring 242 /Aring 243 /ydieresis
244 /Ydieresis 246 /aacute 247 /Aacute 252 /ugrave 253 /Ugrave] def
/Times-Roman /Times-Roman accvec ReEncode
/Times-Italic /Times-Italic accvec ReEncode
/Times-Bold /Times-Bold accvec ReEncode
/Times-BoldItalic /Times-BoldItalic accvec ReEncode
/Helvetica /Helvetica accvec ReEncode
/Helvetica-Oblique /Helvetica-Oblique accvec ReEncode
/Helvetica-Bold /Helvetica-Bold accvec ReEncode
/Helvetica-BoldOblique /Helvetica-BoldOblique  accvec ReEncode
/Courier /Courier accvec ReEncode
/Courier-Oblique /Courier-Oblique accvec ReEncode
/Courier-Bold /Courier-Bold accvec ReEncode
/Courier-BoldOblique /Courier-BoldOblique accvec ReEncode
/oshow {gsave [] 0 sd true charpath stroke gr} def
/stwn { /fs exch def /fn exch def /text exch def fn findfont fs sf
 text sw pop xs add /xs exch def} def
/stwb { /fs exch def /fn exch def /nbas exch def /textf exch def
textf length /tlen exch def nbas tlen gt {/nbas tlendef} if
fn findfont fs sf textf dup length nbas sub nbas getinterval sw
pop neg xs add /xs exch def} def
/accspe [ 65 /plusminus 66 /bar 67 /existential 68 /universal
69 /exclam 70 /numbersign 71 /greater 72 /question 73 /integral
74 /colon 75 /semicolon 76 /less 77 /bracketleft 78 /bracketright
79 /greaterequal 80 /braceleft 81 /braceright 82 /radical
83 /spade 84 /heart 85 /diamond 86 /club 87 /lessequal
88 /multiply 89 /percent 90 /infinity 48 /circlemultiply 49 /circleplus
50 /emptyset 51 /lozenge 52 /bullet 53 /arrowright 54 /arrowup
55 /arrowleft 56 /arrowdown 57 /arrowboth 48 /degree 44 /comma 43 /plus
 45 /angle 42 /angleleft 47 /divide 61 /notequal 40 /equivalence 41 /second
 97 /approxequal 98 /congruent 99 /perpendicular 100 /partialdiff 101 /florin
 102 /intersection 103 /union 104 /propersuperset 105 /reflexsuperset
 106 /notsubset 107 /propersubset 108 /reflexsubset 109 /element 110 /notelement
 111 /gradient 112 /logicaland 113 /logicalor 114 /arrowdblboth
 115 /arrowdblleft 116 /arrowdblup 117 /arrowdblright 118 /arrowdbldown
 119 /ampersand 120 /omega1 121 /similar 122 /aleph ] def
/Symbol /Special accspe ReEncode
gsave .25 .25 scale
%%EndProlog
 gsave 0 0 t 0 0 0 c [] 0 sd 1 lw 5 lw 1 284 m 1 284 l 2 284 l 2 283 l 3 283 l 4
 282 l 5 281 l 5 280 l 6 279 l 7 277 l 8 276 l 8 275 l 9 273 l 10 272 l 11 271 l
 11 269 l 12 268 l 13 267 l 14 265 l 14 264 l 15 263 l 16 263 l 17 262 l 18 261
 l 19 261 l 20 261 l 20 262 l 21 262 l 22 263 l 23 263 l 23 264 l 24 265 l 25
 267 l 26 268 l 27 269 l 27 271 l 28 272 l 29 273 l 30 275 l 30 276 l 31 277 l
 32 279 l 33 280 l 33 281 l 34 282 l 35 283 l 36 283 l 36 284 l 37 284 l 38 284
 l s 38 284 m 39 284 l 40 283 l 41 283 l 42 282 l 42 281 l 43 280 l 44 279 l 45
 277 l 45 276 l 46 275 l 47 273 l 48 272 l 48 271 l 49 269 l 50 268 l 51 267 l
 51 265 l 52 264 l 53 263 l 54 263 l 55 262 l 56 261 l 57 261 l 58 261 l 58 262
 l 59 262 l 60 263 l 61 263 l 61 264 l 62 265 l 63 267 l 64 268 l 64 269 l 65
 271 l 66 272 l 67 273 l 67 275 l 68 276 l 69 277 l 70 279 l 70 280 l 71 281 l
 72 282 l 73 283 l 74 284 l 75 284 l 76 284 l s 76 284 m 76 284 l 77 284 l 78
 283 l 79 283 l 80 282 l 80 281 l 81 280 l 82 279 l 83 277 l 83 276 l 84 275 l
 85 273 l 86 272 l 86 271 l 87 269 l 88 268 l 89 267 l 89 265 l 90 264 l 91 263
 l 92 263 l 92 262 l 93 262 l 94 261 l 95 261 l 96 262 l 97 262 l 98 263 l 99
 264 l 100 265 l 101 267 l 101 268 l 102 269 l 103 271 l 104 272 l 105 273 l 105
 275 l 106 276 l 107 277 l 108 279 l 108 280 l 109 281 l 110 282 l 111 283 l 112
 284 l 113 284 l 114 284 l s 1 lw 114 307 m 111 307 l 108 306 l 105 305 l 103
 304 l 100 302 l 98 301 l 96 298 l 94 296 l 93 294 l 92 291 l 91 288 l 91 285 l
 91 283 l 91 280 l 92 277 l 93 274 l 94 272 l 96 270 l 98 267 l 100 266 l 103
 264 l 105 263 l 108 262 l 111 261 l 114 261 l 116 261 l 119 262 l 122 263 l 125
 264 l 127 266 l 129 267 l 131 270 l 133 272 l 134 274 l 135 277 l 136 280 l 136
 283 l 136 285 l 136 288 l 135 291 l 134 294 l 133 296 l 131 298 l 129 301 l 127
 302 l 125 304 l 122 305 l 119 306 l 116 307 l 114 307 l f 5 lw 281 429 m 281
 429 l 278 427 l 274 426 l 271 424 l 267 423 l 264 422 l 260 422 l 257 422 l 253
 422 l 250 423 l 247 424 l 244 425 l 241 426 l 239 428 l 236 430 l 235 432 l 233
 434 l 232 436 l 231 438 l 231 440 l 230 442 l 231 443 l 231 445 l 232 446 l 233
 448 l 234 448 l 236 449 l 237 449 l 239 449 l 241 449 l 243 448 l 245 447 l 247
 445 l 248 444 l 250 442 l 252 439 l 253 437 l 254 434 l 255 431 l 255 427 l 255
 424 l 255 420 l 254 417 l 254 413 l 252 410 l 251 407 l 249 403 l 247 400 l 244
 397 l 241 395 l s 241 395 m 238 392 l 235 390 l 232 389 l 228 387 l 225 386 l
 221 385 l 218 385 l 214 385 l 211 385 l 207 386 l 204 387 l 201 388 l 198 389 l
 196 391 l 194 393 l 192 395 l 190 397 l 189 399 l 188 401 l 188 403 l 188 405 l
 188 407 l 188 408 l 189 409 l 190 411 l 192 412 l 193 412 l 195 412 l 197 412 l
 198 412 l 200 411 l 202 410 l 204 409 l 206 407 l 207 405 l 209 402 l 210 400 l
 211 397 l 212 394 l 212 390 l 213 387 l 212 384 l 212 380 l 211 376 l 210 373 l
 208 370 l 206 366 l 204 363 l 202 360 l 199 358 l s 199 358 m 196 355 l 193 353
 l 189 352 l 186 350 l 182 349 l 179 348 l 175 348 l 171 348 l 168 348 l 165 349
 l 161 350 l 159 351 l 156 353 l 153 354 l 151 356 l 149 358 l 148 360 l 147 362
 l 146 364 l 145 366 l 145 368 l 145 370 l 146 371 l 147 373 l 148 374 l 149 375
 l 151 375 l 152 375 l 154 375 l 156 375 l 158 374 l 160 373 l 161 372 l 163 370
 l 165 368 l 166 365 l 168 363 l 169 360 l 169 357 l 170 353 l 170 350 l 170 347
 l 169 343 l 168 340 l 167 336 l 166 333 l 164 329 l 161 326 l 159 324 l 156 321
 l s 156 321 m 153 319 l 150 317 l 147 315 l 143 313 l 140 312 l 136 312 l 132
 311 l 129 311 l 125 311 l 122 312 l 119 313 l 116 314 l 113 316 l 111 317 l 109
 319 l 107 321 l 105 323 l 104 325 l 103 327 l 103 329 l 103 331 l 103 333 l 103
 334 l 104 336 l 105 337 l 106 338 l 108 338 l 110 339 l 111 338 l 113 338 l 115
 337 l 117 336 l 119 335 l 121 333 l 122 331 l 124 328 l 125 326 l 126 323 l 127
 320 l 127 317 l 127 313 l 127 310 l 127 306 l 126 303 l 125 299 l 123 296 l 121
 293 l 119 289 l 116 287 l 114 284 l s 116 281 m 116 281 l 119 279 l 121 275 l
 123 272 l 125 269 l 126 265 l 127 262 l 127 258 l 127 255 l 127 251 l 127 248 l
 126 245 l 125 242 l 124 240 l 122 237 l 121 235 l 119 233 l 117 232 l 115 231 l
 113 230 l 111 230 l 110 229 l 108 230 l 106 230 l 105 231 l 104 232 l 103 234 l
 103 235 l 103 237 l 103 239 l 103 241 l 104 243 l 105 245 l 107 247 l 109 249 l
 111 251 l 113 252 l 116 254 l 119 255 l 122 256 l 125 257 l 129 257 l 132 257 l
 136 256 l 140 256 l 143 255 l 147 253 l 150 251 l 153 249 l 156 247 l s 156 247
 m 159 244 l 161 242 l 164 239 l 166 235 l 167 232 l 168 228 l 169 225 l 170 221
 l 170 218 l 170 215 l 169 211 l 169 208 l 168 205 l 166 203 l 165 200 l 163 198
 l 161 196 l 160 195 l 158 194 l 156 193 l 154 193 l 152 193 l 151 193 l 149 193
 l 148 194 l 147 195 l 146 197 l 145 198 l 145 200 l 145 202 l 146 204 l 147 206
 l 148 208 l 149 210 l 151 212 l 153 214 l 156 215 l 159 217 l 161 218 l 165 219
 l 168 220 l 171 220 l 175 220 l 179 220 l 182 219 l 186 218 l 189 216 l 193 215
 l 196 213 l 199 210 l s 199 210 m 202 208 l 204 205 l 206 202 l 208 198 l 210
 195 l 211 192 l 212 188 l 212 184 l 213 181 l 212 178 l 212 174 l 211 171 l 210
 168 l 209 166 l 207 163 l 206 161 l 204 159 l 202 158 l 200 157 l 198 156 l 197
 156 l 195 156 l 193 156 l 192 156 l 190 157 l 189 159 l 188 160 l 188 161 l 188
 163 l 188 165 l 188 167 l 189 169 l 190 171 l 192 173 l 194 175 l 196 177 l 198
 179 l 201 180 l 204 181 l 207 182 l 211 183 l 214 183 l 218 183 l 221 183 l 225
 182 l 228 181 l 232 179 l 235 178 l 238 176 l 241 173 l s 241 173 m 244 171 l
 247 168 l 249 165 l 251 161 l 252 158 l 254 155 l 254 151 l 255 148 l 255 144 l
 255 141 l 255 137 l 254 134 l 253 131 l 252 129 l 250 126 l 248 124 l 247 123 l
 245 121 l 243 120 l 241 119 l 239 119 l 237 119 l 236 119 l 234 120 l 233 120 l
 232 122 l 231 123 l 231 125 l 230 126 l 231 128 l 231 130 l 232 132 l 233 134 l
 235 136 l 236 138 l 239 140 l 241 142 l 244 143 l 247 144 l 250 145 l 253 146 l
 257 146 l 260 146 l 264 146 l 267 145 l 271 144 l 274 142 l 278 141 l 281 139 l
 284 136 l s 284 432 m 454 284 l s 284 136 m 454 284 l s 284 432 m 284 136 l s
 456 284 m 456 284 l 457 284 l 458 283 l 459 283 l 460 282 l 461 281 l 462 280 l
 463 279 l 465 277 l 466 276 l 467 275 l 468 273 l 469 272 l 470 271 l 471 269 l
 473 268 l 474 267 l 475 265 l 476 264 l 477 263 l 478 263 l 479 262 l 481 262 l
 482 261 l 483 261 l 484 261 l 485 262 l 486 262 l 487 263 l 488 263 l 490 264 l
 491 265 l 492 267 l 493 268 l 494 269 l 495 271 l 496 272 l 498 273 l 499 275 l
 500 276 l 501 277 l 502 279 l 503 280 l 504 281 l 506 282 l 507 283 l 508 283 l
 509 284 l 510 284 l 511 284 l s 511 284 m 512 284 l 513 284 l 515 283 l 516 283
 l 517 282 l 518 281 l 519 280 l 520 279 l 521 277 l 523 276 l 524 275 l 525 273
 l 526 272 l 527 271 l 528 269 l 529 268 l 531 267 l 532 265 l 533 264 l 534 263
 l 535 263 l 536 262 l 537 262 l 538 261 l 540 261 l 541 261 l 542 262 l 543 262
 l 544 263 l 545 263 l 546 264 l 548 265 l 549 267 l 550 268 l 551 269 l 552 271
 l 553 272 l 554 273 l 556 275 l 557 276 l 558 277 l 559 279 l 560 280 l 561 281
 l 562 282 l 563 283 l 565 283 l 566 284 l 567 284 l 568 284 l s 1 lw 270 102 m
 270 57 l s 270 102 m 296 102 l s 302 68 m 302 45 l s 302 56 m 306 60 l 308 61 l
 311 61 l 313 60 l 314 56 l 314 45 l s 323 63 m 323 64 l 324 66 l 325 67 l 327
 68 l 332 68 l 334 67 l 335 66 l 336 64 l 336 62 l 335 60 l 333 56 l 322 45 l
 337 45 l s
gr gr
end
%%EOF
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% ghar.eps %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%!PS-Adobe-2.0 EPSF-2.0
%%BoundingBox: 0 0 142 142
%%Title: /usr/users/ak/PAW/SINGLET/ghar.eps
%%Creator: HIGZ Version 1.16/11
%%CreationDate: 93/07/16   10.56
%%EndComments
80 dict begin
/s {stroke} def /l {lineto} def /m {moveto} def /t {translate} def
/sw {stringwidth} def /r {rotate} def /rl {roll} def
/d {rlineto} def /rm {rmoveto} def /gr {grestore} def /f {eofill} def
/c {setrgbcolor} def /lw {setlinewidth} def /sd {setdash} def
/cl {closepath} def /sf {scalefont setfont} def
/box {m dup 0 exch d exch 0 d 0 exch neg d cl} def
/bl {box s} def /bf {box f} def
/mp {newpath /y exch def /x exch def} def
/side {[w .77 mul w .23 mul] .385 w mul sd w 0 l currentpoint t -144 r} def
/mr {mp x y w2 0 360 arc} def /m24 {mr s} def /m20 {mr f} def
/mb {mp x y w2 add m w2 neg 0 d 0 w neg d w 0 d 0 w d cl} def
/mt {mp x y w2 add m w2 neg w neg d w 0 d cl} def
/m21 {mb f} def /m25 {mb s} def /m22 {mt f} def /m26{mt s} def
/m23 {mp x y w2 sub m w2 w d w neg 0 d cl f} def
 /m27 {mp x y w2 add m w3 neg w2 neg d w3 w2 neg d w3 w2 d cl s} def
 /m28 {mp x w2 sub y w2 sub w3 add m w3 0 d 0 w3 neg d w3 0 d 0 w3 d w3 0 d
 0 w3 d w3 neg 0 d 0 w3 d w3 neg 0 d 0 w3 neg d w3 neg 0 d cl s } def
 /m29 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 4 {side} repeat cl fill gr} def
 /m30 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 5 {side} repeat s gr} def /m31 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d
 x w2 sub y w2 add m w w neg d x w2 sub y w2
 sub m w w d s} def
/m2 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d s} def
/m5 {mp x w2 sub y w2 sub m w w d x w2 sub y w2 add m w w neg d s} def
/reencdict 24 dict def /ReEncode {reencdict begin /nco&na exch def
/nfnam exch def /basefontname exch def /basefontdict basefontname findfont def
/newfont basefontdict maxlength dict def basefontdict {exch dup /FID ne
{dup /Encoding eq {exch dup length array copy newfont 3 1 roll put} {exch
newfont 3 1 roll put} ifelse} {pop pop} ifelse } forall newfont
/FontName nfnam put nco&na aload pop nco&na length 2 idiv {newfont
/Encoding get 3 1 roll put} repeat nfnam newfont definefont pop end } def
/accvec [ 176 /agrave 181 /Agrave 190 /acircumflex 192 /Acircumflex
201 /adieresis 204 /Adieresis 209 /ccedilla 210 /Ccedilla 211 /eacute
212 /Eacute 213 /egrave 214 /Egrave 215 /ecircumflex 216 /Ecircumflex
217 /edieresis 218 /Edieresis 219 /icircumflex 220 /Icircumflex
221 /idieresis 222 /Idieresis 223 /ntilde 224 /Ntilde 226 /ocircumflex
228 /Ocircumflex 229 /odieresis 230 /Odieresis 231 /ucircumflex 236 /Ucircumflex
237 /udieresis 238 /Udieresis 239 /aring 242 /Aring 243 /ydieresis
244 /Ydieresis 246 /aacute 247 /Aacute 252 /ugrave 253 /Ugrave] def
/Times-Roman /Times-Roman accvec ReEncode
/Times-Italic /Times-Italic accvec ReEncode
/Times-Bold /Times-Bold accvec ReEncode
/Times-BoldItalic /Times-BoldItalic accvec ReEncode
/Helvetica /Helvetica accvec ReEncode
/Helvetica-Oblique /Helvetica-Oblique accvec ReEncode
/Helvetica-Bold /Helvetica-Bold accvec ReEncode
/Helvetica-BoldOblique /Helvetica-BoldOblique  accvec ReEncode
/Courier /Courier accvec ReEncode
/Courier-Oblique /Courier-Oblique accvec ReEncode
/Courier-Bold /Courier-Bold accvec ReEncode
/Courier-BoldOblique /Courier-BoldOblique accvec ReEncode
/oshow {gsave [] 0 sd true charpath stroke gr} def
/stwn { /fs exch def /fn exch def /text exch def fn findfont fs sf
 text sw pop xs add /xs exch def} def
/stwb { /fs exch def /fn exch def /nbas exch def /textf exch def
textf length /tlen exch def nbas tlen gt {/nbas tlendef} if
fn findfont fs sf textf dup length nbas sub nbas getinterval sw
pop neg xs add /xs exch def} def
/accspe [ 65 /plusminus 66 /bar 67 /existential 68 /universal
69 /exclam 70 /numbersign 71 /greater 72 /question 73 /integral
74 /colon 75 /semicolon 76 /less 77 /bracketleft 78 /bracketright
79 /greaterequal 80 /braceleft 81 /braceright 82 /radical
83 /spade 84 /heart 85 /diamond 86 /club 87 /lessequal
88 /multiply 89 /percent 90 /infinity 48 /circlemultiply 49 /circleplus
50 /emptyset 51 /lozenge 52 /bullet 53 /arrowright 54 /arrowup
55 /arrowleft 56 /arrowdown 57 /arrowboth 48 /degree 44 /comma 43 /plus
 45 /angle 42 /angleleft 47 /divide 61 /notequal 40 /equivalence 41 /second
 97 /approxequal 98 /congruent 99 /perpendicular 100 /partialdiff 101 /florin
 102 /intersection 103 /union 104 /propersuperset 105 /reflexsuperset
 106 /notsubset 107 /propersubset 108 /reflexsubset 109 /element 110 /notelement
 111 /gradient 112 /logicaland 113 /logicalor 114 /arrowdblboth
 115 /arrowdblleft 116 /arrowdblup 117 /arrowdblright 118 /arrowdbldown
 119 /ampersand 120 /omega1 121 /similar 122 /aleph ] def
/Symbol /Special accspe ReEncode
gsave .25 .25 scale
%%EndProlog
 gsave 0 0 t 0 0 0 c [] 0 sd 1 lw 5 lw 1 284 m 1 284 l 2 284 l 2 283 l 3 283 l 4
 282 l 5 281 l 6 280 l 6 279 l 7 277 l 8 276 l 9 275 l 10 273 l 10 272 l 11 271
 l 12 269 l 13 268 l 14 267 l 14 265 l 15 264 l 16 263 l 17 263 l 17 262 l 18
 262 l 19 261 l 20 261 l 21 261 l 21 262 l 22 262 l 23 263 l 24 263 l 25 264 l
 25 265 l 26 267 l 27 268 l 28 269 l 29 271 l 29 272 l 30 273 l 31 275 l 32 276
 l 33 277 l 33 279 l 34 280 l 35 281 l 36 282 l 37 283 l 38 284 l 39 284 l 40
 284 l s 40 284 m 41 284 l 42 283 l 43 283 l 44 282 l 45 281 l 45 280 l 46 279 l
 47 277 l 48 276 l 49 275 l 49 273 l 50 272 l 51 271 l 52 269 l 52 268 l 53 267
 l 54 265 l 55 264 l 56 263 l 57 262 l 58 262 l 59 261 l 60 261 l 61 262 l 62
 262 l 63 263 l 64 263 l 64 264 l 65 265 l 66 267 l 67 268 l 68 269 l 68 271 l
 69 272 l 70 273 l 71 275 l 72 276 l 72 277 l 73 279 l 74 280 l 75 281 l 76 282
 l 76 283 l 77 283 l 78 284 l 79 284 l 80 284 l s 20 lw 80 284 m 193 432 l s 80
 284 m 193 136 l s 193 432 m 193 136 l s 5 lw 80 284 m 148 373 l s 1 lw 124 342
 m 90 325 l 117 304 l cl s 5 lw 193 432 m 91 299 l s 1 lw 132 352 m 166 368 l
 139 389 l cl s 5 lw 197 432 m 197 432 l 200 431 l 204 430 l 207 429 l 210 427 l
 213 426 l 216 423 l 218 421 l 221 419 l 223 416 l 224 413 l 225 410 l 226 408 l
 227 405 l 227 402 l 227 399 l 227 397 l 226 394 l 225 392 l 224 391 l 222 389 l
 221 388 l 219 387 l 218 386 l 216 386 l 214 386 l 212 387 l 211 388 l 209 389 l
 208 391 l 207 392 l 206 394 l 205 397 l 205 399 l 205 402 l 205 405 l 205 408 l
 206 410 l 208 413 l 209 416 l 211 419 l 213 421 l 216 423 l 218 426 l 221 427 l
 225 429 l 228 430 l 231 431 l 235 432 l 239 432 l s 239 432 m 242 432 l 246 431
 l 249 430 l 252 429 l 256 427 l 259 426 l 261 423 l 264 421 l 266 419 l 268 416
 l 270 413 l 271 410 l 272 408 l 272 405 l 272 402 l 272 399 l 272 397 l 271 394
 l 270 392 l 269 391 l 268 389 l 266 388 l 265 387 l 263 386 l 261 386 l 260 386
 l 258 387 l 256 388 l 255 389 l 253 391 l 252 392 l 251 394 l 251 397 l 250 399
 l 250 402 l 250 405 l 251 408 l 252 410 l 253 413 l 255 416 l 257 419 l 259 421
 l 261 423 l 264 426 l 267 427 l 270 429 l 273 430 l 277 431 l 280 432 l 284 432
 l s 284 432 m 288 432 l 291 431 l 295 430 l 298 429 l 301 427 l 304 426 l 307
 423 l 309 421 l 311 419 l 313 416 l 315 413 l 316 410 l 317 408 l 318 405 l 318
 402 l 318 399 l 317 397 l 317 394 l 316 392 l 315 391 l 313 389 l 312 388 l 310
 387 l 308 386 l 307 386 l 305 386 l 303 387 l 302 388 l 300 389 l 299 391 l 298
 392 l 297 394 l 296 397 l 296 399 l 296 402 l 296 405 l 296 408 l 297 410 l 298
 413 l 300 416 l 302 419 l 304 421 l 307 423 l 309 426 l 312 427 l 316 429 l 319
 430 l 322 431 l 326 432 l 329 432 l s 329 432 m 333 432 l 337 431 l 340 430 l
 343 429 l 347 427 l 350 426 l 352 423 l 355 421 l 357 419 l 359 416 l 360 413 l
 362 410 l 363 408 l 363 405 l 363 402 l 363 399 l 363 397 l 362 394 l 361 392 l
 360 391 l 359 389 l 357 388 l 356 387 l 354 386 l 352 386 l 350 386 l 349 387 l
 347 388 l 346 389 l 344 391 l 343 392 l 342 394 l 341 397 l 341 399 l 341 402 l
 341 405 l 342 408 l 343 410 l 344 413 l 345 416 l 347 419 l 350 421 l 352 423 l
 355 426 l 358 427 l 361 429 l 364 430 l 368 431 l 371 432 l 375 432 l s 371 136
 m 371 136 l 368 137 l 364 138 l 361 139 l 358 141 l 355 142 l 352 145 l 350 147
 l 347 149 l 345 152 l 344 155 l 343 158 l 342 160 l 341 163 l 341 166 l 341 169
 l 341 171 l 342 174 l 343 176 l 344 177 l 346 179 l 347 180 l 349 181 l 350 182
 l 352 182 l 354 182 l 356 181 l 357 180 l 359 179 l 360 177 l 361 176 l 362 174
 l 363 171 l 363 169 l 363 166 l 363 163 l 363 160 l 362 158 l 360 155 l 359 152
 l 357 149 l 355 147 l 352 145 l 350 142 l 347 141 l 343 139 l 340 138 l 337 137
 l 333 136 l 329 136 l s 329 136 m 326 136 l 322 137 l 319 138 l 316 139 l 312
 141 l 309 142 l 307 145 l 304 147 l 302 149 l 300 152 l 298 155 l 297 158 l 296
 160 l 296 163 l 296 166 l 296 169 l 296 171 l 297 174 l 298 176 l 299 177 l 300
 179 l 302 180 l 303 181 l 305 182 l 307 182 l 308 182 l 310 181 l 312 180 l 313
 179 l 315 177 l 316 176 l 317 174 l 317 171 l 318 169 l 318 166 l 318 163 l 317
 160 l 316 158 l 315 155 l 313 152 l 311 149 l 309 147 l 307 145 l 304 142 l 301
 141 l 298 139 l 295 138 l 291 137 l 288 136 l 284 136 l s 284 136 m 280 136 l
 277 137 l 273 138 l 270 139 l 267 141 l 264 142 l 261 145 l 259 147 l 257 149 l
 255 152 l 253 155 l 252 158 l 251 160 l 250 163 l 250 166 l 250 169 l 251 171 l
 251 174 l 252 176 l 253 177 l 255 179 l 256 180 l 258 181 l 260 182 l 261 182 l
 263 182 l 265 181 l 266 180 l 268 179 l 269 177 l 270 176 l 271 174 l 272 171 l
 272 169 l 272 166 l 272 163 l 272 160 l 271 158 l 270 155 l 268 152 l 266 149 l
 264 147 l 261 145 l 259 142 l 256 141 l 252 139 l 249 138 l 246 137 l 242 136 l
 239 136 l s 239 136 m 235 136 l 231 137 l 228 138 l 225 139 l 221 141 l 218 142
 l 216 145 l 213 147 l 211 149 l 209 152 l 208 155 l 206 158 l 205 160 l 205 163
 l 205 166 l 205 169 l 205 171 l 206 174 l 207 176 l 208 177 l 209 179 l 211 180
 l 212 181 l 214 182 l 216 182 l 218 182 l 219 181 l 221 180 l 222 179 l 224 177
 l 225 176 l 226 174 l 227 171 l 227 169 l 227 166 l 227 163 l 226 160 l 225 158
 l 224 155 l 223 152 l 221 149 l 218 147 l 216 145 l 213 142 l 210 141 l 207 139
 l 204 138 l 200 137 l 197 136 l 193 136 l s 375 432 m 375 136 l s 375 432 m 488
 284 l s 375 136 m 488 284 l s 489 284 m 489 284 l 490 284 l 491 283 l 492 283 l
 492 282 l 493 281 l 494 280 l 495 279 l 496 277 l 496 276 l 497 275 l 498 273 l
 499 272 l 500 271 l 500 269 l 501 268 l 502 267 l 503 265 l 504 264 l 504 263 l
 505 263 l 506 262 l 507 262 l 508 261 l 509 261 l 510 262 l 511 262 l 512 263 l
 513 264 l 514 265 l 515 267 l 516 268 l 516 269 l 517 271 l 518 272 l 519 273 l
 519 275 l 520 276 l 521 277 l 522 279 l 523 280 l 523 281 l 524 282 l 525 283 l
 526 283 l 527 284 l 528 284 l s 528 284 m 529 284 l 530 284 l 531 283 l 532 282
 l 533 281 l 534 280 l 535 279 l 535 277 l 536 276 l 537 275 l 538 273 l 539 272
 l 539 271 l 540 269 l 541 268 l 542 267 l 543 265 l 543 264 l 544 263 l 545 263
 l 546 262 l 547 262 l 547 261 l 548 261 l 549 261 l 550 262 l 551 262 l 551 263
 l 552 263 l 553 264 l 554 265 l 554 267 l 555 268 l 556 269 l 557 271 l 558 272
 l 558 273 l 559 275 l 560 276 l 561 277 l 562 279 l 562 280 l 563 281 l 564 282
 l 565 283 l 566 283 l 566 284 l 567 284 l 568 284 l s 1 lw 273 84 m 271 82 l
 269 78 l 267 74 l 266 69 l 266 64 l 267 59 l 269 55 l 271 51 l 273 49 l s 281
 80 m 281 57 l s 281 69 m 283 71 l 285 72 l 288 72 l 290 71 l 293 69 l 294 65 l
 294 63 l 293 60 l 290 58 l 288 57 l 285 57 l 283 58 l 281 60 l s 300 84 m 302
 82 l 305 78 l 307 74 l 308 69 l 308 64 l 307 59 l 305 55 l 302 51 l 300 49 l s
gr gr
end
%%EOF
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% ct1.eps %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%!PS-Adobe-2.0 EPSF-2.0
%%BoundingBox: 0 0 142 142
%%Title: /usr/users/ak/PAW/SINGLET/ct1.eps
%%Creator: HIGZ Version 1.16/11
%%CreationDate: 93/10/03   12.44
%%EndComments
80 dict begin
/s {stroke} def /l {lineto} def /m {moveto} def /t {translate} def
/sw {stringwidth} def /r {rotate} def /rl {roll} def
/d {rlineto} def /rm {rmoveto} def /gr {grestore} def /f {eofill} def
/c {setrgbcolor} def /lw {setlinewidth} def /sd {setdash} def
/cl {closepath} def /sf {scalefont setfont} def
/box {m dup 0 exch d exch 0 d 0 exch neg d cl} def
/bl {box s} def /bf {box f} def
/mp {newpath /y exch def /x exch def} def
/side {[w .77 mul w .23 mul] .385 w mul sd w 0 l currentpoint t -144 r} def
/mr {mp x y w2 0 360 arc} def /m24 {mr s} def /m20 {mr f} def
/mb {mp x y w2 add m w2 neg 0 d 0 w neg d w 0 d 0 w d cl} def
/mt {mp x y w2 add m w2 neg w neg d w 0 d cl} def
/m21 {mb f} def /m25 {mb s} def /m22 {mt f} def /m26{mt s} def
/m23 {mp x y w2 sub m w2 w d w neg 0 d cl f} def
 /m27 {mp x y w2 add m w3 neg w2 neg d w3 w2 neg d w3 w2 d cl s} def
 /m28 {mp x w2 sub y w2 sub w3 add m w3 0 d 0 w3 neg d w3 0 d 0 w3 d w3 0 d
 0 w3 d w3 neg 0 d 0 w3 d w3 neg 0 d 0 w3 neg d w3 neg 0 d cl s } def
 /m29 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 4 {side} repeat cl fill gr} def
 /m30 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 5 {side} repeat s gr} def /m31 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d
 x w2 sub y w2 add m w w neg d x w2 sub y w2
 sub m w w d s} def
/m2 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d s} def
/m5 {mp x w2 sub y w2 sub m w w d x w2 sub y w2 add m w w neg d s} def
/reencdict 24 dict def /ReEncode {reencdict begin /nco&na exch def
/nfnam exch def /basefontname exch def /basefontdict basefontname findfont def
/newfont basefontdict maxlength dict def basefontdict {exch dup /FID ne
{dup /Encoding eq {exch dup length array copy newfont 3 1 roll put} {exch
newfont 3 1 roll put} ifelse} {pop pop} ifelse } forall newfont
/FontName nfnam put nco&na aload pop nco&na length 2 idiv {newfont
/Encoding get 3 1 roll put} repeat nfnam newfont definefont pop end } def
/accvec [ 176 /agrave 181 /Agrave 190 /acircumflex 192 /Acircumflex
201 /adieresis 204 /Adieresis 209 /ccedilla 210 /Ccedilla 211 /eacute
212 /Eacute 213 /egrave 214 /Egrave 215 /ecircumflex 216 /Ecircumflex
217 /edieresis 218 /Edieresis 219 /icircumflex 220 /Icircumflex
221 /idieresis 222 /Idieresis 223 /ntilde 224 /Ntilde 226 /ocircumflex
228 /Ocircumflex 229 /odieresis 230 /Odieresis 231 /ucircumflex 236 /Ucircumflex
237 /udieresis 238 /Udieresis 239 /aring 242 /Aring 243 /ydieresis
244 /Ydieresis 246 /aacute 247 /Aacute 252 /ugrave 253 /Ugrave] def
/Times-Roman /Times-Roman accvec ReEncode
/Times-Italic /Times-Italic accvec ReEncode
/Times-Bold /Times-Bold accvec ReEncode
/Times-BoldItalic /Times-BoldItalic accvec ReEncode
/Helvetica /Helvetica accvec ReEncode
/Helvetica-Oblique /Helvetica-Oblique accvec ReEncode
/Helvetica-Bold /Helvetica-Bold accvec ReEncode
/Helvetica-BoldOblique /Helvetica-BoldOblique  accvec ReEncode
/Courier /Courier accvec ReEncode
/Courier-Oblique /Courier-Oblique accvec ReEncode
/Courier-Bold /Courier-Bold accvec ReEncode
/Courier-BoldOblique /Courier-BoldOblique accvec ReEncode
/oshow {gsave [] 0 sd true charpath stroke gr} def
/stwn { /fs exch def /fn exch def /text exch def fn findfont fs sf
 text sw pop xs add /xs exch def} def
/stwb { /fs exch def /fn exch def /nbas exch def /textf exch def
textf length /tlen exch def nbas tlen gt {/nbas tlendef} if
fn findfont fs sf textf dup length nbas sub nbas getinterval sw
pop neg xs add /xs exch def} def
/accspe [ 65 /plusminus 66 /bar 67 /existential 68 /universal
69 /exclam 70 /numbersign 71 /greater 72 /question 73 /integral
74 /colon 75 /semicolon 76 /less 77 /bracketleft 78 /bracketright
79 /greaterequal 80 /braceleft 81 /braceright 82 /radical
83 /spade 84 /heart 85 /diamond 86 /club 87 /lessequal
88 /multiply 89 /percent 90 /infinity 48 /circlemultiply 49 /circleplus
50 /emptyset 51 /lozenge 52 /bullet 53 /arrowright 54 /arrowup
55 /arrowleft 56 /arrowdown 57 /arrowboth 48 /degree 44 /comma 43 /plus
 45 /angle 42 /angleleft 47 /divide 61 /notequal 40 /equivalence 41 /second
 97 /approxequal 98 /congruent 99 /perpendicular 100 /partialdiff 101 /florin
 102 /intersection 103 /union 104 /propersuperset 105 /reflexsuperset
 106 /notsubset 107 /propersubset 108 /reflexsubset 109 /element 110 /notelement
 111 /gradient 112 /logicaland 113 /logicalor 114 /arrowdblboth
 115 /arrowdblleft 116 /arrowdblup 117 /arrowdblright 118 /arrowdbldown
 119 /ampersand 120 /omega1 121 /similar 122 /aleph ] def
/Symbol /Special accspe ReEncode
gsave .25 .25 scale
%%EndProlog
 gsave 0 0 t 0 0 0 c [] 0 sd 1 lw 80 293 m 78 293 l 77 293 l 76 292 l 75 292 l
 74 291 l 73 291 l 73 290 l 72 289 l 71 288 l 71 287 l 71 286 l 70 285 l 70 283
 l 71 282 l 71 281 l 71 280 l 72 279 l 73 278 l 73 277 l 74 277 l 75 276 l 76
 276 l 77 275 l 78 275 l 80 275 l 81 275 l 82 275 l 83 276 l 84 276 l 85 277 l
 86 277 l 87 278 l 87 279 l 88 280 l 88 281 l 88 282 l 89 283 l 89 285 l 88 286
 l 88 287 l 88 288 l 87 289 l 87 290 l 86 291 l 85 291 l 84 292 l 83 292 l 82
 293 l 81 293 l 80 293 l f 20 lw 80 284 m 193 432 l s 80 284 m 193 136 l s 193
 432 m 193 136 l s 5 lw 197 432 m 197 432 l 200 431 l 204 430 l 207 429 l 210
 427 l 213 426 l 216 423 l 218 421 l 221 419 l 223 416 l 224 413 l 225 410 l 226
 408 l 227 405 l 227 402 l 227 399 l 227 397 l 226 394 l 225 392 l 224 391 l 222
 389 l 221 388 l 219 387 l 218 386 l 216 386 l 214 386 l 212 387 l 211 388 l 209
 389 l 208 391 l 207 392 l 206 394 l 205 397 l 205 399 l 205 402 l 205 405 l 205
 408 l 206 410 l 208 413 l 209 416 l 211 419 l 213 421 l 216 423 l 218 426 l 221
 427 l 225 429 l 228 430 l 231 431 l 235 432 l 239 432 l s 239 432 m 242 432 l
 246 431 l 249 430 l 252 429 l 256 427 l 259 426 l 261 423 l 264 421 l 266 419 l
 268 416 l 270 413 l 271 410 l 272 408 l 272 405 l 272 402 l 272 399 l 272 397 l
 271 394 l 270 392 l 269 391 l 268 389 l 266 388 l 265 387 l 263 386 l 261 386 l
 260 386 l 258 387 l 256 388 l 255 389 l 253 391 l 252 392 l 251 394 l 251 397 l
 250 399 l 250 402 l 250 405 l 251 408 l 252 410 l 253 413 l 255 416 l 257 419 l
 259 421 l 261 423 l 264 426 l 267 427 l 270 429 l 273 430 l 277 431 l 280 432 l
 284 432 l s 284 432 m 288 432 l 291 431 l 295 430 l 298 429 l 301 427 l 304 426
 l 307 423 l 309 421 l 311 419 l 313 416 l 315 413 l 316 410 l 317 408 l 318 405
 l 318 402 l 318 399 l 317 397 l 317 394 l 316 392 l 315 391 l 313 389 l 312 388
 l 310 387 l 308 386 l 307 386 l 305 386 l 303 387 l 302 388 l 300 389 l 299 391
 l 298 392 l 297 394 l 296 397 l 296 399 l 296 402 l 296 405 l 296 408 l 297 410
 l 298 413 l 300 416 l 302 419 l 304 421 l 307 423 l 309 426 l 312 427 l 316 429
 l 319 430 l 322 431 l 326 432 l 329 432 l s 329 432 m 333 432 l 337 431 l 340
 430 l 343 429 l 347 427 l 350 426 l 352 423 l 355 421 l 357 419 l 359 416 l 360
 413 l 362 410 l 363 408 l 363 405 l 363 402 l 363 399 l 363 397 l 362 394 l 361
 392 l 360 391 l 359 389 l 357 388 l 356 387 l 354 386 l 352 386 l 350 386 l 349
 387 l 347 388 l 346 389 l 344 391 l 343 392 l 342 394 l 341 397 l 341 399 l 341
 402 l 341 405 l 342 408 l 343 410 l 344 413 l 345 416 l 347 419 l 350 421 l 352
 423 l 355 426 l 358 427 l 361 429 l 364 430 l 368 431 l 371 432 l 375 432 l s
 371 136 m 371 136 l 368 137 l 364 138 l 361 139 l 358 141 l 355 142 l 352 145 l
 350 147 l 347 149 l 345 152 l 344 155 l 343 158 l 342 160 l 341 163 l 341 166 l
 341 169 l 341 171 l 342 174 l 343 176 l 344 177 l 346 179 l 347 180 l 349 181 l
 350 182 l 352 182 l 354 182 l 356 181 l 357 180 l 359 179 l 360 177 l 361 176 l
 362 174 l 363 171 l 363 169 l 363 166 l 363 163 l 363 160 l 362 158 l 360 155 l
 359 152 l 357 149 l 355 147 l 352 145 l 350 142 l 347 141 l 343 139 l 340 138 l
 337 137 l 333 136 l 329 136 l s 329 136 m 326 136 l 322 137 l 319 138 l 316 139
 l 312 141 l 309 142 l 307 145 l 304 147 l 302 149 l 300 152 l 298 155 l 297 158
 l 296 160 l 296 163 l 296 166 l 296 169 l 296 171 l 297 174 l 298 176 l 299 177
 l 300 179 l 302 180 l 303 181 l 305 182 l 307 182 l 308 182 l 310 181 l 312 180
 l 313 179 l 315 177 l 316 176 l 317 174 l 317 171 l 318 169 l 318 166 l 318 163
 l 317 160 l 316 158 l 315 155 l 313 152 l 311 149 l 309 147 l 307 145 l 304 142
 l 301 141 l 298 139 l 295 138 l 291 137 l 288 136 l 284 136 l s 284 136 m 280
 136 l 277 137 l 273 138 l 270 139 l 267 141 l 264 142 l 261 145 l 259 147 l 257
 149 l 255 152 l 253 155 l 252 158 l 251 160 l 250 163 l 250 166 l 250 169 l 251
 171 l 251 174 l 252 176 l 253 177 l 255 179 l 256 180 l 258 181 l 260 182 l 261
 182 l 263 182 l 265 181 l 266 180 l 268 179 l 269 177 l 270 176 l 271 174 l 272
 171 l 272 169 l 272 166 l 272 163 l 272 160 l 271 158 l 270 155 l 268 152 l 266
 149 l 264 147 l 261 145 l 259 142 l 256 141 l 252 139 l 249 138 l 246 137 l 242
 136 l 239 136 l s 239 136 m 235 136 l 231 137 l 228 138 l 225 139 l 221 141 l
 218 142 l 216 145 l 213 147 l 211 149 l 209 152 l 208 155 l 206 158 l 205 160 l
 205 163 l 205 166 l 205 169 l 205 171 l 206 174 l 207 176 l 208 177 l 209 179 l
 211 180 l 212 181 l 214 182 l 216 182 l 218 182 l 219 181 l 221 180 l 222 179 l
 224 177 l 225 176 l 226 174 l 227 171 l 227 169 l 227 166 l 227 163 l 226 160 l
 225 158 l 224 155 l 223 152 l 221 149 l 218 147 l 216 145 l 213 142 l 210 141 l
 207 139 l 204 138 l 200 137 l 197 136 l 193 136 l s 375 432 m 375 136 l s 375
 432 m 432 432 l s 375 136 m 432 136 l s 1 lw 375 441 m 374 441 l 373 440 l 372
 440 l 371 440 l 370 439 l 369 438 l 368 437 l 367 437 l 367 436 l 366 434 l 366
 433 l 366 432 l 366 431 l 366 430 l 366 429 l 367 428 l 367 427 l 368 426 l 369
 425 l 370 424 l 371 424 l 372 423 l 373 423 l 374 423 l 375 423 l 376 423 l 377
 423 l 378 423 l 379 424 l 380 424 l 381 425 l 382 426 l 383 427 l 383 428 l 384
 429 l 384 430 l 384 431 l 384 432 l 384 433 l 384 434 l 383 436 l 383 437 l 382
 437 l 381 438 l 380 439 l 379 440 l 378 440 l 377 440 l 376 441 l 375 441 l f
 375 145 m 374 145 l 373 145 l 372 145 l 371 144 l 370 144 l 369 143 l 368 142 l
 367 141 l 367 140 l 366 139 l 366 138 l 366 137 l 366 136 l 366 135 l 366 134 l
 367 132 l 367 131 l 368 131 l 369 130 l 370 129 l 371 128 l 372 128 l 373 128 l
 374 127 l 375 127 l 376 127 l 377 128 l 378 128 l 379 128 l 380 129 l 381 130 l
 382 131 l 383 131 l 383 132 l 384 134 l 384 135 l 384 136 l 384 137 l 384 138 l
 384 139 l 383 140 l 383 141 l 382 142 l 381 143 l 380 144 l 379 144 l 378 145 l
 377 145 l 376 145 l 375 145 l f 278 91 m 275 96 l 271 100 l 267 102 l 258 102 l
 254 100 l 249 96 l 247 91 l 245 85 l 245 74 l 247 68 l 249 63 l 254 59 l 258 57
 l 267 57 l 271 59 l 275 63 l 278 68 l s 290 64 m 293 65 l 296 68 l 296 45 l s
 318 118 m 315 116 l 313 113 l 311 108 l 310 103 l 310 98 l 311 93 l 313 89 l
 315 85 l 318 83 l s 326 114 m 326 95 l 327 92 l 329 91 l 332 91 l s 323 106 m
 331 106 l s 337 118 m 339 116 l 341 113 l 344 108 l 345 103 l 345 98 l 344 93 l
 341 89 l 339 85 l 337 83 l s
gr gr
end
%%EOF
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% ct2.eps %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%!PS-Adobe-2.0 EPSF-2.0
%%BoundingBox: 0 0 142 142
%%Title: /usr/users/ak/PAW/SINGLET/ct2.eps
%%Creator: HIGZ Version 1.16/11
%%CreationDate: 93/10/03   12.45
%%EndComments
80 dict begin
/s {stroke} def /l {lineto} def /m {moveto} def /t {translate} def
/sw {stringwidth} def /r {rotate} def /rl {roll} def
/d {rlineto} def /rm {rmoveto} def /gr {grestore} def /f {eofill} def
/c {setrgbcolor} def /lw {setlinewidth} def /sd {setdash} def
/cl {closepath} def /sf {scalefont setfont} def
/box {m dup 0 exch d exch 0 d 0 exch neg d cl} def
/bl {box s} def /bf {box f} def
/mp {newpath /y exch def /x exch def} def
/side {[w .77 mul w .23 mul] .385 w mul sd w 0 l currentpoint t -144 r} def
/mr {mp x y w2 0 360 arc} def /m24 {mr s} def /m20 {mr f} def
/mb {mp x y w2 add m w2 neg 0 d 0 w neg d w 0 d 0 w d cl} def
/mt {mp x y w2 add m w2 neg w neg d w 0 d cl} def
/m21 {mb f} def /m25 {mb s} def /m22 {mt f} def /m26{mt s} def
/m23 {mp x y w2 sub m w2 w d w neg 0 d cl f} def
 /m27 {mp x y w2 add m w3 neg w2 neg d w3 w2 neg d w3 w2 d cl s} def
 /m28 {mp x w2 sub y w2 sub w3 add m w3 0 d 0 w3 neg d w3 0 d 0 w3 d w3 0 d
 0 w3 d w3 neg 0 d 0 w3 d w3 neg 0 d 0 w3 neg d w3 neg 0 d cl s } def
 /m29 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 4 {side} repeat cl fill gr} def
 /m30 {mp gsave x w2 sub y w2 add w3 sub m currentpoint t
 5 {side} repeat s gr} def /m31 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d
 x w2 sub y w2 add m w w neg d x w2 sub y w2
 sub m w w d s} def
/m2 {mp x y w2 sub m 0 w d x w2 sub y m w 0 d s} def
/m5 {mp x w2 sub y w2 sub m w w d x w2 sub y w2 add m w w neg d s} def
/reencdict 24 dict def /ReEncode {reencdict begin /nco&na exch def
/nfnam exch def /basefontname exch def /basefontdict basefontname findfont def
/newfont basefontdict maxlength dict def basefontdict {exch dup /FID ne
{dup /Encoding eq {exch dup length array copy newfont 3 1 roll put} {exch
newfont 3 1 roll put} ifelse} {pop pop} ifelse } forall newfont
/FontName nfnam put nco&na aload pop nco&na length 2 idiv {newfont
/Encoding get 3 1 roll put} repeat nfnam newfont definefont pop end } def
/accvec [ 176 /agrave 181 /Agrave 190 /acircumflex 192 /Acircumflex
201 /adieresis 204 /Adieresis 209 /ccedilla 210 /Ccedilla 211 /eacute
212 /Eacute 213 /egrave 214 /Egrave 215 /ecircumflex 216 /Ecircumflex
217 /edieresis 218 /Edieresis 219 /icircumflex 220 /Icircumflex
221 /idieresis 222 /Idieresis 223 /ntilde 224 /Ntilde 226 /ocircumflex
228 /Ocircumflex 229 /odieresis 230 /Odieresis 231 /ucircumflex 236 /Ucircumflex
237 /udieresis 238 /Udieresis 239 /aring 242 /Aring 243 /ydieresis
244 /Ydieresis 246 /aacute 247 /Aacute 252 /ugrave 253 /Ugrave] def
/Times-Roman /Times-Roman accvec ReEncode
/Times-Italic /Times-Italic accvec ReEncode
/Times-Bold /Times-Bold accvec ReEncode
/Times-BoldItalic /Times-BoldItalic accvec ReEncode
/Helvetica /Helvetica accvec ReEncode
/Helvetica-Oblique /Helvetica-Oblique accvec ReEncode
/Helvetica-Bold /Helvetica-Bold accvec ReEncode
/Helvetica-BoldOblique /Helvetica-BoldOblique  accvec ReEncode
/Courier /Courier accvec ReEncode
/Courier-Oblique /Courier-Oblique accvec ReEncode
/Courier-Bold /Courier-Bold accvec ReEncode
/Courier-BoldOblique /Courier-BoldOblique accvec ReEncode
/oshow {gsave [] 0 sd true charpath stroke gr} def
/stwn { /fs exch def /fn exch def /text exch def fn findfont fs sf
 text sw pop xs add /xs exch def} def
/stwb { /fs exch def /fn exch def /nbas exch def /textf exch def
textf length /tlen exch def nbas tlen gt {/nbas tlendef} if
fn findfont fs sf textf dup length nbas sub nbas getinterval sw
pop neg xs add /xs exch def} def
/accspe [ 65 /plusminus 66 /bar 67 /existential 68 /universal
69 /exclam 70 /numbersign 71 /greater 72 /question 73 /integral
74 /colon 75 /semicolon 76 /less 77 /bracketleft 78 /bracketright
79 /greaterequal 80 /braceleft 81 /braceright 82 /radical
83 /spade 84 /heart 85 /diamond 86 /club 87 /lessequal
88 /multiply 89 /percent 90 /infinity 48 /circlemultiply 49 /circleplus
50 /emptyset 51 /lozenge 52 /bullet 53 /arrowright 54 /arrowup
55 /arrowleft 56 /arrowdown 57 /arrowboth 48 /degree 44 /comma 43 /plus
 45 /angle 42 /angleleft 47 /divide 61 /notequal 40 /equivalence 41 /second
 97 /approxequal 98 /congruent 99 /perpendicular 100 /partialdiff 101 /florin
 102 /intersection 103 /union 104 /propersuperset 105 /reflexsuperset
 106 /notsubset 107 /propersubset 108 /reflexsubset 109 /element 110 /notelement
 111 /gradient 112 /logicaland 113 /logicalor 114 /arrowdblboth
 115 /arrowdblleft 116 /arrowdblup 117 /arrowdblright 118 /arrowdbldown
 119 /ampersand 120 /omega1 121 /similar 122 /aleph ] def
/Symbol /Special accspe ReEncode
gsave .25 .25 scale
%%EndProlog
 gsave 0 0 t 0 0 0 c [] 0 sd 1 lw 80 293 m 78 293 l 77 293 l 76 292 l 75 292 l
 74 291 l 73 291 l 73 290 l 72 289 l 71 288 l 71 287 l 71 286 l 70 285 l 70 283
 l 71 282 l 71 281 l 71 280 l 72 279 l 73 278 l 73 277 l 74 277 l 75 276 l 76
 276 l 77 275 l 78 275 l 80 275 l 81 275 l 82 275 l 83 276 l 84 276 l 85 277 l
 86 277 l 87 278 l 87 279 l 88 280 l 88 281 l 88 282 l 89 283 l 89 285 l 88 286
 l 88 287 l 88 288 l 87 289 l 87 290 l 86 291 l 85 291 l 84 292 l 83 292 l 82
 293 l 81 293 l 80 293 l f 20 lw 80 284 m 193 432 l s 80 284 m 193 136 l s 193
 432 m 193 136 l s 1 lw 193 441 m 192 441 l 191 440 l 190 440 l 189 440 l 188
 439 l 187 438 l 186 437 l 185 437 l 185 436 l 184 434 l 184 433 l 184 432 l 184
 431 l 184 430 l 184 429 l 185 428 l 185 427 l 186 426 l 187 425 l 188 424 l 189
 424 l 190 423 l 191 423 l 192 423 l 193 423 l 194 423 l 195 423 l 196 423 l 197
 424 l 198 424 l 199 425 l 200 426 l 201 427 l 201 428 l 202 429 l 202 430 l 202
 431 l 202 432 l 202 433 l 202 434 l 201 436 l 201 437 l 200 437 l 199 438 l 198
 439 l 197 440 l 196 440 l 195 440 l 194 441 l 193 441 l f 193 145 m 192 145 l
 191 145 l 190 145 l 189 144 l 188 144 l 187 143 l 186 142 l 185 141 l 185 140 l
 184 139 l 184 138 l 184 137 l 184 136 l 184 135 l 184 134 l 185 132 l 185 131 l
 186 131 l 187 130 l 188 129 l 189 128 l 190 128 l 191 128 l 192 127 l 193 127 l
 194 127 l 195 128 l 196 128 l 197 128 l 198 129 l 199 130 l 200 131 l 201 131 l
 201 132 l 202 134 l 202 135 l 202 136 l 202 137 l 202 138 l 202 139 l 201 140 l
 201 141 l 200 142 l 199 143 l 198 144 l 197 144 l 196 145 l 195 145 l 194 145 l
 193 145 l f 5 lw 197 432 m 197 432 l 200 431 l 204 430 l 207 429 l 210 427 l
 213 426 l 216 423 l 218 421 l 221 419 l 223 416 l 224 413 l 225 410 l 226 408 l
 227 405 l 227 402 l 227 399 l 227 397 l 226 394 l 225 392 l 224 391 l 222 389 l
 221 388 l 219 387 l 218 386 l 216 386 l 214 386 l 212 387 l 211 388 l 209 389 l
 208 391 l 207 392 l 206 394 l 205 397 l 205 399 l 205 402 l 205 405 l 205 408 l
 206 410 l 208 413 l 209 416 l 211 419 l 213 421 l 216 423 l 218 426 l 221 427 l
 225 429 l 228 430 l 231 431 l 235 432 l 239 432 l s 239 432 m 242 432 l 246 431
 l 249 430 l 252 429 l 256 427 l 259 426 l 261 423 l 264 421 l 266 419 l 268 416
 l 270 413 l 271 410 l 272 408 l 272 405 l 272 402 l 272 399 l 272 397 l 271 394
 l 270 392 l 269 391 l 268 389 l 266 388 l 265 387 l 263 386 l 261 386 l 260 386
 l 258 387 l 256 388 l 255 389 l 253 391 l 252 392 l 251 394 l 251 397 l 250 399
 l 250 402 l 250 405 l 251 408 l 252 410 l 253 413 l 255 416 l 257 419 l 259 421
 l 261 423 l 264 426 l 267 427 l 270 429 l 273 430 l 277 431 l 280 432 l 284 432
 l s 280 136 m 280 136 l 277 137 l 273 138 l 270 139 l 267 141 l 264 142 l 261
 145 l 259 147 l 257 149 l 255 152 l 253 155 l 252 158 l 251 160 l 250 163 l 250
 166 l 250 169 l 251 171 l 251 174 l 252 176 l 253 177 l 255 179 l 256 180 l 258
 181 l 260 182 l 261 182 l 263 182 l 265 181 l 266 180 l 268 179 l 269 177 l 270
 176 l 271 174 l 272 171 l 272 169 l 272 166 l 272 163 l 272 160 l 271 158 l 270
 155 l 268 152 l 266 149 l 264 147 l 261 145 l 259 142 l 256 141 l 252 139 l 249
 138 l 246 137 l 242 136 l 239 136 l s 239 136 m 235 136 l 231 137 l 228 138 l
 225 139 l 221 141 l 218 142 l 216 145 l 213 147 l 211 149 l 209 152 l 208 155 l
 206 158 l 205 160 l 205 163 l 205 166 l 205 169 l 205 171 l 206 174 l 207 176 l
 208 177 l 209 179 l 211 180 l 212 181 l 214 182 l 216 182 l 218 182 l 219 181 l
 221 180 l 222 179 l 224 177 l 225 176 l 226 174 l 227 171 l 227 169 l 227 166 l
 227 163 l 226 160 l 225 158 l 224 155 l 223 152 l 221 149 l 218 147 l 216 145 l
 213 142 l 210 141 l 207 139 l 204 138 l 200 137 l 197 136 l 193 136 l s 1 lw
 175 91 m 173 96 l 169 100 l 164 102 l 156 102 l 151 100 l 147 96 l 145 91 l 143
 85 l 143 74 l 145 68 l 147 63 l 151 59 l 156 57 l 164 57 l 169 59 l 173 63 l
 175 68 l s 186 63 m 186 64 l 187 66 l 188 67 l 190 68 l 195 68 l 197 67 l 198
 66 l 199 64 l 199 62 l 198 60 l 196 56 l 185 45 l 200 45 l s 215 118 m 213 116
 l 211 113 l 209 108 l 208 103 l 208 98 l 209 93 l 211 89 l 213 85 l 215 83 l s
 224 114 m 224 95 l 225 92 l 227 91 l 229 91 l s 221 106 m 228 106 l s 235 118 m
 237 116 l 239 113 l 241 108 l 242 103 l 242 98 l 241 93 l 239 89 l 237 85 l 235
 83 l s
gr gr
end
%%EOF



